



The following constitutes the ruling of the court and has the force and effect therein described.

Signed October 16, 2018

Mark X. Mullin United States Bankruptcy Judge

IN THE UNITED STATES BANKRUPTCY COURT FOR THE NORTHERN DISTRICT OF TEXAS

FORT WORTH DIVISION

In re: § Chapter 11

§

PREFERRED CARE INC., et al., § Case No.: 17-44642

§

Debtors. § Jointly Administered

§

ORDER (A) GRANTING AUTHORITY TO: (I) TRANSFER THE OPERATIONS AND RELATED ASSETS OF 5 NEW MEXICO FACILITIES FREE AND CLEAR OF ALL LIENS, CLAIMS, ENCUMBRANCES, AND INTERESTS; (II) ASSUME AND ASSIGN CERTAIN EXECUTORY CONTRACTS AND UNEXPIRED LEASES; AND (III) REJECT AND TERMINATE THE REAL PROPERTY LEASES; (B) APPROVING THE FORM OF OPERATIONS TRANSFER AGREEMENT; AND (C) APPROVING THE TRANSITION AGREEMENT

This matter has come before the Court on the Motion for Order (A) Granting Authority to (I) Transfer the Operations and Related Assets of 5 New Mexico Facilities Free and Clear of All Liens, Claims, Encumbrances, and Interests, (II) Assume and Assign Certain Executory Contracts and Unexpired Leases, and; (III) Reject and Terminate the Real Property Leases; (B)

Approving the Form of Operations Transfer Agreement; and (C) Approving the Transition Agreement [Docket No. 1077] (the "Motion"). The Motion was filed by five (5) limited partnerships (the "Omega 5 Debtors" or "Movants") which operate five (5) skilled nursing homes in New Mexico (the "Omega 5 Facilities"). The Omega 5 Debtors are debtors and debtors-in-possession in the bankruptcy cases jointly administered in Case No. 17-44642. The Movants have requested, pursuant to sections 363, 365, and 1146 of Title 11 of the United States Code (the "Bankruptcy Code") and Rules 2002, 6004, 6006, and 9014 of the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), the entry of this order (the "Sale Order"):

- (a) authorizing the sale and transfer of the operations of the Omega 5 Facilities and assets necessary for the operations of the Omega 5 Facilities (defined more fully in the OTAs as the "Assets"), free and clear of liens, claims, encumbrances, and interests;
- (b) approving the *Operations Transfer Agreements* (each individually an "OTA" and collectively, the "**OTAs**"), substantially in the form attached hereto as **Exhibit A**, by and between the Omega 5 Debtors and the Purchaser of the Omega 5 Facilities;
- (c) approving the assumption and assignment of the Assumed Contracts;
- (d) approving the rejection and termination of the real property leases associated with each of the Omega 5 Facilities (the "Lease Terminations"); and
- (e) approving the Transition Agreement and First Amendment to Transition Agreement (collectively, the "**Transition Agreement**") between the Debtors and Omega; and
- (f) granting related relief, including relief related to the proposed Cure Costs associated with the Assumed Contracts.

-

Capitalized terms not defined herein shall carry the meanings ascribed to them in the Motion or the OTAs, as applicable.

On October 10, 2018, the Court conducted a hearing on the Motion (the "Sale Hearing"), and, having considered: (a) the Motion and the relief requested therein and transactions contemplated thereby; (b) all objections, if any, to the Motion; and (c) all matters brought to the Court's attention at the Sale Hearing, including the arguments of counsel; and having taken judicial notice of the materials on file in the above-enumerated Chapter 11 Cases, the Court has determined that the Movants have established just cause for the relief granted herein. Accordingly;

THE COURT HEREBY FINDS AND DETERMINES THAT:²

- 1. The Court has jurisdiction over the Motion pursuant to 28 U.S.C. §§ 157 and 1334. This matter is a core proceeding under 28 U.S.C. § 157(b)(2). Venue is proper in this Court under 28 U.S.C. §§ 1408 and 1409.
- 2. The statutory predicates for the relief sought in the Motion are sections 105(a), 363(b), (f), and (m), 365, and 1146(c) of the Bankruptcy Code and Bankruptcy Rules 2002, 6004, 6006 and 9014.
- 3. The Omega 5 Debtors have Medicare Provider Agreements with the U.S. Department of Health and Human Services' Centers for Medicare & Medicaid Services ("CMS") (collectively, the "Provider Agreements"). The Omega 5 Debtors are subject to Medicare statutes, regulations, procedures, policies, and rules.

3

This Sale Order constitutes the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052. All findings of fact and conclusions of law announced by the Court at the hearing shall be deemed additional findings of fact and conclusions of law in this matter and are incorporated herein. When appropriate, all findings of fact shall be construed as conclusions of law, and all conclusions of law shall be construed as findings of fact.

- 4. This Sale Order constitutes a final order within the meaning of 28 U.S.C. § 158(a). Notwithstanding the provisions of Bankruptcy Rules 6004(h) and 6006(d), the Court expressly finds that there is no just cause for delay and that the Sale Order shall be effective and enforceable immediately upon entry based upon the representations and facts presented.
- 5. The Motion has been served on the parties listed on the *Certificate of Service* filed at Docket No. 1079, as well as all parties having filed a Notice of Appearance in the Chapter 11 Cases and/or registered to receive electronic service. Notice of the Sale Hearing was provided to the parties listed on the *Certificates of Service* filed at Docket Nos. 1079 and 1114.
- 6. The Court finds that the scope and manner of the service provided by the Movants was proper, timely, adequate, and sufficient, in accordance with Bankruptcy Code §§ 363 and 365, Bankruptcy Rules 2002, 2002(i), 6004, 6006, and 9014, and in compliance with all prior orders of the Court related to the Assets, including the *Order Granting Motion for Order Establishing Notice Procedures and Approving Form Notice of Commencement of Cases* [Docket No. 61] (the "Notice Procedures Order"). No further notice of the Motion or the Sale Hearing was required. The Court further finds that a reasonable opportunity to object or to be heard regarding the relief requested in the Motion has been afforded to all creditors and parties in interest.
- 7. The Movants have the full power and authority to execute the Transition Agreement and OTAs and all other documents that are: 1) referenced in or contemplated by the Transition Agreement and OTAs or 2) necessary or appropriate to effectuate the transfers of the Omega 5 Debtors' operations and Assets. All actions contemplated by the Transition Agreement and OTAs have been duly and validly authorized, and the Movants have the full power and authority to consummate the transactions contemplated by the Transition Agreement and OTA.

No further consents or approvals, other than entry of this Sale Order, are required for the Movants to consummate the Transition Agreement and OTAs, except as specifically provided for in this Sale Order.

- 8. The relief sought by the Movants in the Motion, including, but not limited to, the authorization of the sale and transfer of the operations and Assets of the Omega 5 Facilities, approval of the Transition Agreement and OTAs, approval of the assumption and assignment of the Assumed Contracts; and approval of the Lease Terminations, is, at this time, in the best interests of the Movants, their bankruptcy estates, their creditors and their interest holders.
- 9. The Movants have demonstrated both (i) good, sufficient, and sound business purpose and justification for the subject transaction and (ii) compelling circumstances for approval of the Transition Agreement and OTAs pursuant to Bankruptcy Code §§ 363(b), and (f) and 365.
- 10. The OTAs were proposed and entered into by the Movants and the Purchaser in good faith. The consideration set forth in the OTAs constitutes fair value for the Assets. Neither the Movants nor the Purchaser have engaged in any conduct that would cause or permit the OTAs to be avoided under Bankruptcy Code § 363(n). The Purchaser is not an "insider" of the Omega 5 Debtors, as that term is defined in Bankruptcy Code § 101. The Purchaser has no connections, is not affiliates or an insider of, nor does it have any undisclosed agreements with either the Omega 5 Debtors, or any of their insiders or affiliates.
- 11. The Purchaser is a good faith transferee of the Assets under Bankruptcy Code § 363(m) and, as such, is entitled to all of the protections afforded thereby. The Purchaser has acted in good faith within the meaning of Bankruptcy Code § 363(m) and will rely on entry of this Sale Order and this good faith determination in consummating and closing the transactions.

- 12. The OTAs was not entered into for the purpose of hindering, delaying or defrauding creditors under the Bankruptcy Code or under the laws of the United States, any state, territory, possession or the District of Columbia.
- 13. The consideration provided for the Assets (i) is fair and reasonable, (ii) constitutes reasonably equivalent value and fair consideration under the Bankruptcy Code and applicable non-bankruptcy law, and (iii) is the highest and best value available under the circumstances.
- 14. Omega affiliates, the lessors under leases with the Omega 5 Debtors, would not have agreed to terminate the leases with the Omega 5 Debtors and to release claims against the Movants if the Purchaser had not agreed to the OTAs and entered into new leases for the Omega 5 Facilities. The Purchaser would not have agreed to the OTAs if it was not to be made pursuant to § 363(f) and §365 of the Bankruptcy Code, free and clear of all claims, liens encumbrances and other interests of any kind or nature whatsoever or if the Purchaser would, or in the future could, be liable for any of the claims, liens, encumbrances or other interests against the Omega 5 Debtors or the Assets. Pursuant to § 363(f) of the Bankruptcy Code, the Purchaser is entitled to know that the Assets are not infected with latent claims that will be asserted against the Purchaser after the Closing.
- Assets to the Purchaser will be a legal, valid and effective transfer of the Assets and will vest the Purchaser with all right, title and interest of the Omega 5 Debtors in the Assets, free and clear of any and all liens (statutory or otherwise), pledges, mortgages, deeds of trust, security interests, claims, reclamation claims, charges, hypothecations, assignments, licenses, liabilities, beneficial interests, options, rights of first or last refusal, options to purchase, priority or other security agreements or preferential arrangements of any kind or nature, rights of rescission (statutory or

otherwise), causes of action, covenants, restrictions, easements, rights of setoff, defects in title or other encumbrances of any kind, whether legal or equitable, secured or unsecured, material or immaterial, contingent or non-contingent, arising out of or in connection with or in any way related to the Omega 5 Debtors, the Assets, or the operation of the business of the Omega 5 Debtors prior to Closing, including any claim as defined in § 101(5) of the Bankruptcy Code, whether arising prior to or subsequent to the commencement of this case (but not on or subsequent to the Closing Date), and whether imposed by agreement, law, equity or otherwise, including, without limitation, any claims arising under doctrines of successor liability with respect to the Assets (collectively, the "Claims"), except as specifically provided in paragraphs 37-41 of this Sale Order. The Claims are hereinafter referred to as the "Interests."

- 16. The Movants are authorized to transfer the Assets free and clear of the Interests because one or more of the standards set forth in Bankruptcy Code § 363(f) has been satisfied with respect to each such interest. Any objection of a secured creditor or other holder of an Interest that timely objected to the transfer of the Assets is overruled, as one or more of the other subsections of Bankruptcy Code § 363(f) is met with respect to such party.
- 17. The Assumed Contracts are an integral part of the Assets and are included in the OTA. All Cure Costs, if any, under the Assumed Contracts will be paid by the Omega 5 Debtors within 30 days of the entry of this Sale Order. The Purchaser has demonstrated adequate assurance of future performance under the Assumed Contracts.
- 18. The Closing under the OTAs is conditioned upon entry of this Sale Order, and the assumption and assignment of the Provider Agreements is subject to the Medicare statutes, regulations, procedures, policies, and rules.

NOW, THEREFORE, IT IS HEREBY ORDERED, ADJUDGED AND DECREED THAT:

- 19. The Motion is GRANTED as set forth herein and the transfer of the Assets and the assumption and assignment of the Assumed Contracts to the Purchaser is hereby authorized as set forth in this Sale Order.
- 20. All objections to the Motion, if any, that have not been withdrawn, waived, or settled, and all reservations of rights included therein, are hereby overruled on the merits.

A. Approval of the Transition Agreement and Operations Transfer Agreements

- 21. The Transition Agreement and OTAs, all exhibits and schedules thereto, and all of the terms and conditions thereof are hereby approved.
- 22. Pursuant to Bankruptcy Code §§ 363(b), and (f), and 365, the Movants, as debtors-in-possession, are authorized to consummate the transfer of the Assets, pursuant to and in accordance with the terms and conditions of the OTAs, including, without limitation, to convey the Assets to the Purchaser and to assume and assign the Assumed Contracts to the Purchaser. The terms of the OTAs are approved in all respects.
- 23. Without the need for any additional order of this Court, the Movants and their employees and agents are authorized to execute and deliver, and empowered to perform under, consummate and implement the Transition Agreement and OTAs, together with all additional instruments and documents that may be reasonably necessary or desirable to implement the transactions, and to take all further actions as may be reasonably requested by the Purchaser or otherwise required under the Transition Agreement and OTAs, including without limitation the requirements set forth in Sections 2.5, 2.9, and 2.21 of the OTAs related to administration of accounts receivable (the "Accounts Receivable Procedures").

24. Without the need for any additional order of this Court, the Debtors are authorized and directed to comply with the provisions of the Accounts Receivable Procedures set forth in Sections 2.5, 2.9, and 2.21 of the OTAs, including without limitation, recognition of the ownership interest of Purchaser in funds that are proceeds of Purchaser's accounts receivable generated after the Closing Date, and the transfer of the amount of such funds to Purchaser in accordance with the Accounts Receivable Procedures.

B. Approval of the Lease Terminations

- 25. The Lease Terminations are hereby approved pursuant to 11 U.S.C. § 365. The leases associated with each of the Omega 5 Facilities shall be rejected and terminated as of the Closing Date.
- 26. With respect to each lease applicable to the Omega 5 Debtors, the Motion constitutes a separate contested matter as contemplated by Bankruptcy Rule 9014. This Sale Order shall be deemed a separate order with respect to the rejection and termination of each such lease associated with the Omega 5 Facilities.

C. Transfer of the Assets Free and Clear of Interests

27. The Omega 5 Debtors are authorized to transfer the Assets to Purchaser in accordance with the terms of the OTAs. The Omega 5 Debtors shall transfer the Assets to the Purchaser on the Closing Date, in accordance with the OTAs, and such transfer shall constitute a legal, valid, binding and effective transfer of the Assets and shall vest the Purchaser with good title and all right, title and interest in the Assets in accordance with the OTAs free and clear of all Interests, except as specifically provided in paragraphs 37-41 of this Sale Order. The Assets shall include all of the Omega 5 Debtors' Transferred Records, which shall remain with the applicable Omega 5 Facility and be owned by the Purchaser on and after the Closing Date.

- 28. Pursuant to §§ 105(a) and 363(f) of the Bankruptcy Code, the transfer of the Assets to the Purchaser shall be, and hereby is, free and clear of any and all Interests, rights and encumbrances whatsoever, except as may otherwise be set forth explicitly in the OTAs or this Sale Order.
- 29. Except as expressly permitted by the OTAs or this Sale Order, all persons and entities holding or asserting Interests arising under or out of, in connection with, or in any way relating to the Omega 5 Debtors, the Assets, the operation of the Omega 5 Debtors' businesses prior to the Closing Date or the transfer of the Assets to the Purchaser, are hereby forever barred, estopped and permanently enjoined from asserting such Interests against the Purchaser or any of the Assets. All persons and entities are hereby forever prohibited and enjoined from taking any action that would adversely affect or interfere with the ability of the Omega 5 Debtors to transfer the Assets to the Purchaser in accordance with the terms of the OTAs and this Sale Order.
- 30. This Sale Order is and shall be effective as a determination that, upon transfer of the Assets to the Purchaser, pursuant to the OTAs, all Interests in, against or relating to any of the Assets conveyed to the Purchaser have been and hereby are terminated and declared to be unconditionally released, discharged and terminated, except as specifically provided in paragraphs 37-41 of this Sale Order. This Sale Order is and shall be binding upon and shall govern the acts of all persons and entities, including, without limitation, all filing agents, filing officers, title agents, title companies, insurance companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal and local officials, and all other persons and entities who may be required by operation of law, the duties of their office, or contract, to accept file, register or otherwise record or release any documents or instruments, or who may be required to report or insure any title or

state of title in or to any of the Assets; and each of the foregoing persons and entities is hereby directed to accept for filing any and all of the documents and instruments necessary and appropriate to consummate the transactions contemplated by the OTAs, except as provided in paragraphs 37-41 of this Sale Order.

D. Assumption and Assignment of Assumed Contracts

31. The Omega 5 Debtors are hereby authorized and directed, pursuant to Bankruptcy Code §§ 105(a), 363 and 365 to assume, then transfer and assign the Assumed Contracts (as defined more fully in the OTA) to the Purchaser, subject to the terms of the OTAs, which shall vest the Purchaser with all right, title and interest in and to the Assumed Contracts, free and clear of all Interests which have, or could have been, asserted by the Omega 5 Debtors or their creditors in connection with their bankruptcy cases, except as specifically provided in paragraphs 37-41 of this Sale Order.

32. [INTENTIONALLY LEFT BLANK]

- 33. The Purchaser and the Omega 5 Debtors have provided adequate assurance of future performance under the Assumed Contracts and the proposed assumption and assignment of these contracts and leases satisfies the requirements of the Bankruptcy Code, including, *inter alia*, §§ 365(b)(1) and (3) and 365(f), to the extent applicable. Any counterparty to any Assumed Contract that did not object to the proposed assumption and assignment of its agreement is hereby deemed to have consented to the assumption and assignment contemplated herein and any Cure Costs proposed by the Omega 5 Debtors.
- 34. On the Closing Date, the Assumed Contracts will be assigned to the Purchaser, subject to the terms of the OTAs, and will remain valid and binding and in full force and effect in accordance with their respective terms for the benefit of the Purchaser, notwithstanding any

provision in such contracts or leases, or under applicable law (including those described in Bankruptcy Code §§ 365(b)(2), (f)(1), and (3)) that prohibits, restricts, or conditions such assignment or transfer, or that terminates or modifies, or permits a party other than the Omega 5 Debtors to terminate or modify such contracts or leases on account of such assignment or transfer pursuant to Bankruptcy Code § 365(f), except as specifically provided in paragraphs 37-41 of this Sale Order.

- 35. Except as otherwise provided herein, pursuant to § 365(k), upon Closing and payment of the Cure Costs by the Debtors, the Debtors are relieved of any liability for any claims of any kind arising from any of the Assumed Contracts assigned to the Purchaser, and any counterparty to any such contract or lease is hereby forever barred and estopped from asserting any default existing prior to the Closing Date against the Omega 5 Debtors, their estates, or the Purchaser, except as specifically provided in paragraphs 37-41 of this Sale Order. For the avoidance of doubt, nothing in this Sale Order releases the Omega 5 Debtors' liabilities pursuant to their New Mexico Medicaid Provider Participation Agreements.
- 36. The Omega 5 Debtors are further authorized to undertake any and all actions necessary or appropriate to consummate the proposed assignment of the Assumed Contracts to the Purchaser, as specified in the Motion and the OTA.

E. Additional Provisions

37. Notwithstanding anything in this Sale Order or the OTAs, the Omega 5 Debtors shall assume the Provider Agreements and shall assign the relevant Provider Agreement(s) to the Purchaser, effective on the Closing Date and subject to the Omega 5 Debtors' payments to the United States of America (the "United States") as specified below.

- 38. Notwithstanding anything in this Sale Order or the OTAs, the Provider Agreements shall be automatically assigned to the Purchaser upon a change in ownership pursuant to 42 C.F.R. § 489.18(c), and upon assignment, the Provider Agreements shall be subject to all applicable Medicare and/or Medicaid statutes, regulations, policies, procedures and rules, and shall be subject to the terms and conditions under which the Provider Agreements were originally issued, including, but not limited to, the repayment of all pre-assignment Medicare overpayments and all other monetary liabilities, regardless of whether yet determined by CMS.. The Provider Agreements and Purchaser shall be subject to compliance with applicable health and safety standards pursuant to all Medicare and/or Medicaid statutes, regulations, policies, procedures and rules.
- 39. Notwithstanding anything in this Sale Order or the OTAs, the Omega 5 Debtors and the Purchaser shall submit all cost reports pursuant to all Medicare statutes, regulations, policies, procedures and rules. Should the Omega 5 Debtors fail to comply with their obligations of this paragraph, the United States shall be entitled to suspend payments to the Purchaser under the applicable Provider Agreement in accordance with Medicare statutes, regulations, policies, procedures and rules until such time as the required cost reports are filed by the Omega 5 Debtors or the Purchaser. The Omega 5 Debtors shall cooperate with the Purchaser in the filing of the required cost reports.
- 40. Nothing in this Sale Order as so ordered by the Court or the OTAs shall relieve or be construed to relieve the Omega 5 Debtors or any Purchaser from complying with all Medicare and/or New Mexico State statutes, regulations, policies, procedures and rules, including, but not limited to, the requirement that the Omega 5 Debtors and any Purchaser apply for and obtain CMS approval of a change of ownership by the filing of Form CMS-855A.

- 41. Nothing in this Sale Order or the OTAs shall constitute a compromise or waiver by the United States of its ability to take any affirmative or defensive action in these bankruptcy proceedings that is not inconsistent with this Sale Order. The United States shall retain its authority under the Medicare statutes, regulations, procedures, policies and rules to review, approve, deny, or pay Medicare claims made by the Omega 5 Debtors or any Purchaser in the ordinary course of business. Nothing in this Sale Order or the OTAs shall impair or affect the United States' right, claim, defense or ability to recoup, set off, or otherwise recover Medicare overpayments or other monetary liabilities from the Omega 5 Debtors or from any Purchaser in accordance with the Medicare statutes, regulations, procedures, policies and rules.
- 42. The transactions are undertaken by the Purchaser in good faith, as that term is used in Bankruptcy Code § 363(m). Accordingly, the reversal or modification of the authorization provided herein to consummate the transfer shall not affect the validity of the sale of the Assets to the Purchaser, unless such authorization is duly stayed prior to the Closing Date. The Purchaser is a transferee in good faith of the Assets and, upon the Closing Date, is entitled to all of the protections afforded by Bankruptcy Code § 363(m).
- 43. The failure to specifically reference any particular provision of the OTAs in this Sale Order shall not diminish or impair the effectiveness of such provision, it being the intent of this Court that the OTAs be authorized and approved in their entirety.
- 44. The Transition Agreement and OTAs and any related agreements, documents or other instruments may be modified, amended or supplemented by the parties thereto, in a writing signed by both parties, and in accordance with the terms thereof, without further order of this Court, provided that such modification, amendment, or supplement shall not have a material adverse effect on the Movants' bankruptcy estates.

- 45. Notwithstanding anything in this Sale Order, the Motion, the OTAs, the Transition Agreement, any asset purchase agreement, any list or schedule of assumed contracts, or assumed and assigned contracts or cure amounts, including any Cure Cost Notice, any document related to any of the foregoing, or any other order of the Court (including, without limitation, any other provision that purports to be preemptory or supervening): (i) nothing herein or therein shall permit or otherwise effect a sale, an assignment or any other transfer of (a) any insurance policies that have been issued by Federal Insurance Company, Westchester Surplus Lines Insurance Company, Illinois Union Insurance Company or any of their affiliates, predecessors or successors or that provide coverage at any time to any of the Debtors (or their predecessors) and all agreements, documents or instruments relating thereto (collectively, the "Chubb Insurance Contracts"), and/or (b) any rights, benefits, claims, rights to payments and/or recoveries under such Chubb Insurance Contracts; (ii) the Chubb Insurance Contracts and/or any rights, benefits, claims, rights to payments and/or recoveries under or relating to such Chubb Insurance Contracts (a) are not and shall not be deemed to be Assets, or a portion of any of the assets sold, assigned or otherwise transferred as a part of the sale and transfer contemplated hereby, and (b) shall be deemed to be excluded from the definition of Assumed Contracts under the OTAs; and (iii) nothing therein alters or modifies the terms and conditions of the Chubb Insurance Contracts.
 - 46. The provisions of this Sale Order are non-severable and mutually dependent.
- 47. In the event of any inconsistency between the terms and provisions of this Sale Order and the OTAs, the terms and provisions of this Sale Order shall control. To the extent this Sale Order does not include, or otherwise address, any provision contained in both the Motion and the OTAs and where such provision in the Motion and the OTAs are inconsistent, the OTAs

shall govern. In the event of any inconsistency between the terms and provisions of paragraphs 37-41 of this Sale Order on the one hand, and the terms and provisions of any other paragraph in this Sale Order, the OTAs, and any related transaction documents or exhibits on the other hand, the terms and provisions of paragraphs 37-41 of this Sale Order shall control.

- 48. Subject to paragraphs 37-41 of this Sale Order, this Court shall retain jurisdiction to: (i) enforce and implement the terms and provisions of the OTAs (including any breach of the OTAs), all amendments thereto, any waivers and consents thereunder, and each of the agreements executed in connection therewith in all respects; and (ii) determine (by motion and without the necessity for an adversary proceeding, where appropriate) any proceeding, dispute or controversy arising out of or related to this Sale Order, the OTAs or any related transaction documents.
- 49. Notwithstanding the provisions of Bankruptcy Rules 6004(h) and 6006(d), there is no stay pursuant to Bankruptcy Rule 6004(h) or 6006(d) and this Sale Order shall be effective and enforceable immediately upon entry.

###END OF ORDER###

Respectfully submitted by:

/s/ Stephen A. McCartin

Stephen A. McCartin (TX 13374700) Mark C. Moore (TX 24074751)

FOLEY GARDERE
Foley & Lardner LLP

2021 McKinney Avenue, Suite 1600

Dallas, TX 75201

Telephone: (214) 999-3000 Facsimile: (214) 999-4667 smccartin@foley.com mmoore@foley.com

COUNSEL TO DEBTORS AND DEBTORS-IN-POSSESSION

EXHIBIT A FORM OPERATIONS TRANSFER AGREEMENT

OPERATIONS TRANSFER AGREEMENT

This	OPERATIONS	TRANSFER	AGREEMENT	("Agreement")	is	entered	into	as	of
[], 2018 (" Eff o	ective Date"),	by and among Bl	oomfield Health	Faci	ilities, L.	P. ("E	xisti	ng
Operator"), [_] ("New Operato	r").						

RECITALS

- A. Existing Operator is the named tenant under that certain Amended and Restated Lease Agreement dated as of June 13, 2013 by and between [______] ("Landlord") and Existing Operator (the "Existing Lease"), pursuant to which Existing Operator is the licensed operator of Bloomfield Nursing and Rehabilitation located at 803 Hacienda Lane, Bloomfield, New Mexico, (the "Facility").
- B. Existing Operator now desires to cease operating the Facility effective as of 12:01 a.m. MST on the Closing Date (as hereinafter defined) (the "**Effective Time**").
- C. Pursuant to the terms of this Agreement, the New Operator desires to acquire, and the Existing Operator desires to transfer to the New Operator, the business and operations of the Facility and, in connection therewith, the Assets (as hereinafter defined).
- D. Simultaneously with the Closing, (i) Landlord, New Operator and certain of their respective Affiliates desire to enter into a Master Lease Agreement (the "Genesis-Omega Master Lease") pursuant to which New Operator will lease the Facility from Landlord.
- E. Upon the Closing (i) the Existing Lease shall be terminated with respect to the Facility, (ii) the Genesis-Omega Master Lease shall become effective with respect to the Facility simultaneously therewith and (iii) the transactions contemplated by this Agreement shall be consummated.
- F. Certain capitalized terms used herein have the meanings given to them on Exhibit A hereto.

AGREEMENT

NOW, THEREFORE, in consideration of the mutual promises contained herein, and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto agree that:

SECTION 1

TRANSFER OF ASSETS; DELIVERIES UPON CLOSING DATE; CONDITIONS PRECEDENT TO OBLIGATIONS TO CLOSE

1.1 <u>Transfer of Assets</u>. At the Effective Time, and upon the terms and subject to the conditions of this Agreement, the Existing Operator shall transfer, convey, assign and deliver to the New Operator, and the New Operator shall acquire from the Existing Operator, free and clear of all Liens, other than Permitted Liens, all right, title and interest of the Existing Operator in and to the assets of Existing Operator that are owned or used by Existing Operator in connection with the operations of the Facility including, without limitation, the following (collectively, the "Assets"):

- (a) all furniture, fixtures, furnishings and equipment owned by Existing Operator and used in operations of the Facility, including without limitation, any vehicles used in the operations of the Facility (collectively, the "FF&E");
- (b) all information technology equipment related to the Facility's phone system (including related handsets and voicemail systems), security system, nurse call system and any related system wiring (collectively, the "IT Equipment");
 - (c) the Assumed Contracts (as defined in Section 5.12 below);
- (d) to the extent Existing Operator's interest is assignable and/or transferable pursuant to applicable law and to the extent New Operator in its sole discretion elects to assume the same, all licenses, permits, approvals, certifications, certificates of need (or the equivalent), and Medicare and Medicaid provider numbers, which assumption is addressed further in Section 2.9 below;
- (e) all architects and engineers plans and specifications relating to the Facility (to the extent in possession), telephone and facsimile numbers relating to the Facility (including all "800" numbers), all post office box addresses associated with the Facility and all direct messaging and health information exchange accounts associated with the Facility;
 - (f) the Transferred Records (as defined below); and
- (g) any and all other items of tangible and intangible personal property owned or leased by Existing Operator and used solely in connection with the use, leasing and maintenance of the Facility (collectively, the "Personal Property"), and any goodwill of Existing Operator associated with the business operated at the Facility.
- 1.2 <u>Excluded Assets</u>. Notwithstanding anything to the contrary contained herein, the Assets shall not include the following items (collectively, the "Excluded Assets"):
- (a) Existing Operator's rights under this Agreement or any other agreement between the Existing Operator and New Operator;
- (b) all of Existing Operator's bank accounts, cash, cash equivalents, security deposits, securities and accounts receivable (including third party settlements), prepaid accounts, real estate and insurance escrows and inter-company accounts;
- (c) all refunds for taxes, fees, assessments and charges for the period prior to the Effective Time;
- (d) amounts of any nature which are or might be due to Existing Operator for goods provided, services rendered, or any other transaction of any type prior to the Effective Time;
- (e) refunds, rebates and dividends paid in respect of workers compensation or other insurance premiums paid by Existing Operator prior to the Effective Time, and refunds and additional recoveries by or payments to Existing Operator from any person for services, provision of goods or supplies, or any other transactions prior to the Effective Time;
- (f) all refunds arising out of retrospective premium adjustments under insurance policies covering the Facility or operations thereof relating to the period prior to the Effective Time;

- (g) inventory disposed of in the ordinary course of business prior to the Effective Time;
- (h) any provider tax payments related to periods prior to the Effective Time;
- (i) any and all Medicaid and Medicare audit and case mix appeal rights and payments for periods prior to the Effective Time;
- (j) any state Medicaid or Medicare reimbursements or adjustments for services rendered prior to the Effective Time;
 - (k) the New Operator Excluded Contracts;
 - (l) equipment leased pursuant to any lease that is not an Assumed Contract;
 - (m) Existing Operator's financial records, corporate minute books and seals;
- (n) any insurance proceeds paid or payable to Existing Operator for claims arising prior to the Effective Time;
- (o) all licenses, permits, approvals, certifications, and certificate of need rights that are not assignable or transferable pursuant to applicable law;
- (p) and subject to Section 2.20, all computer software, software licenses, time clocks, computer equipment on pharmacy carts and all computer hardware other than the IT Equipment transferred pursuant to Section 1.1(b);
- (q) all causes of action, including, but not limited to, causes of action under Chapter 5 of the Bankruptcy Code.
- 1.3 <u>Assumed Liabilities</u>. As of the Effective Time, New Operator shall not assume any liabilities or obligations of Existing Operator whatsoever, fixed or contingent, other than liabilities and obligations assumed by New Operator pursuant hereto with respect to the Assumed Contracts and the Assumed ETO to the extent such obligations and liabilities relate to periods after the Effective Time (collectively, the "Assumed Liabilities"). In no event shall the Assumed Liabilities include any obligations of Existing Operator relating to (a) brokerage or finder's fees or commissions owed to any person retained by Existing Operator in connection with the transactions contemplated by this Agreement or (b) third party payor programs, including, without limitation, Medicare or Medicaid, arising, accruing, or relating to the time period on or before the Effective Time.
- Closing. Subject to the terms and conditions of this Agreement, the transfer of the Assets and the operations of the Facility shall take place at a closing (the "Closing") to be held effective 12:01 am on November 1, 2018, subject to satisfaction or waiver of the conditions set forth in Section 1.7 and Section 1.8, at the offices of Williams Mullen located at 200 South 10th Street, Suite 1600, Richmond, Virginia 23219, or at such other date or place as Existing Operator and New Operator may mutually agree upon in writing. Alternatively, the Closing may be effected via the electronic exchange of executed documents and closing deliverables. "Closing Date" shall mean the day on which the Closing occurs. In the event that the conditions set forth in Section 1.7 and Section 1.8 are not satisfied or waived on or prior to November 1, 2018, the Closing shall be effective at 12:01am on the first day of the first month following the satisfaction or waiver of such conditions or on such other date as Existing Operator and New Operator may mutually agree.

- 1.5 <u>Existing Operator's Deliveries Upon Closing Date</u>. Existing Operator shall deliver the following to New Operator on the Closing Date (collectively, the "Existing Operator Documents"):
- (a) a duly executed assignment and assumption agreement substantially in the form attached hereto as Exhibit B (the "Assignment and Assumption Agreement");
 - (b) a duly executed bill of sale, in the form attached hereto as Exhibit C;
- (c) such additional bills of sale and other appropriate instruments of assignment and conveyance, as to the Personal Property, in form mutually but reasonably satisfactory to New Operator and Existing Operator, dated as of the Closing Date, conveying all title to the Assets, including the Personal Property, free and clear of all liens, liabilities, security interests or encumbrances except as otherwise permitted herein;
- (d) assignments of all intangible property used by the Existing Operator in the operation of the Facility including, without limitation, documents, chattel paper, instruments, contract rights, resident trust accounts, good will, going concern value, general intangibles, the right to use the trade names and lists of phone numbers, arising from or in connection with Existing Operator's operation or use of any part of the Assets, and excluding all Excluded Assets;
- (e) [a duly executed non-competition agreement substantially in the form of Exhibit D attached hereto (the "Non-Competition Agreement")] [NOTE: New Mexico Facilities only];
- (f) a then current resident listing which shall include such information for the residents of the Facility as is reasonably acceptable to New Operator;
- (g) to the extent not already delivered by Existing Operator, and to the extent available, originals or copies of all of the (A) Assumed Contracts, and (B) licenses and permits for the Facility, to the extent transferrable and included in the Assets;
- (h) a certificate, in form and substance reasonably satisfactory to New Operator, duly executed by an officer of Existing Operator, dated the Closing Date, annexing and certifying as true and complete (A) a Certificate of Account Status of Existing Operator issued by the Texas Comptroller as of a recent date demonstrating that Existing Operator has paid all franchise taxes in the State of Texas to the extent then due and payable, and (B) evidence of the authority of Existing Operator to execute and deliver the Existing Operator Documents in order to effectuate the Closing;
- (i) [insurance certificates evidencing the insurance required in accordance with Section 2.22;] [NOTE: Arizona Facilities only]
 - (j) the accounting of patient funds required by Section 2.2 hereof; and
 - (k) the duly executed Bond Escrow Agreement, if applicable.
- 1.6 <u>New Operator's Deliveries Upon Closing Date</u>. New Operator shall deliver the following to Existing Operator on the Closing Date (collectively, the "New Operator Documents"):
 - (a) the duly executed Assignment and Assumption Agreement; and
 - (b) the duly executed Non-Competition Agreement;

- (c) the duly executed Bond Escrow Agreement, if applicable.
- 1.7 <u>Conditions Precedent to New Operator's Obligation to Close</u>. The New Operator's obligation to consummate the transactions described in this Agreement, and to take the actions required to be taken by New Operator, at the Closing and following the Effective Time is subject to the satisfaction, at or prior to Closing, of each of the following conditions (any of which may be waived by New Operator, in whole or in part):
- (a) Accuracy of Representations. Each representation and warranty of Existing Operator in this Agreement must be accurate in all material respects as of the Closing Date as if made on the Closing Date, except for those representations and warranties which refer to facts existing at a specific date, which shall be accurate in all material respects as of such date.
- (b) Deliveries. At or prior to Closing, the New Operator shall have received the deliveries required under Section 1.5.
- (c) No Injunction. There must not be in effect any injunction, judgment, order, decree, ruling, or charge under any action, suit or proceeding before any court or quasi-judicial or administrative agency of any federal, state, local, or foreign jurisdiction or before any arbitrator that prohibits transfer of operations of, or the Assets used in connection with, the Facilities.
- (d) Governmental Authorizations. At or prior to the Closing, the New Operator shall have obtained the consents and approvals set forth on <u>Schedule 3.3</u>, which shall include, without limitation, the issuance of new food service permits with respect to the Facility.
- (e) Termination of Existing Lease. At or prior to the Closing, the Existing Operator shall have delivered to the New Operator a duly executed termination agreement, terminating the Existing Lease as it relates to the Facility.
- (f) Genesis-Omega Master Lease. At the Closing, Landlord and New Operator shall have entered into the Genesis-Omega Master Lease.
- (g) Performance of Covenants. Existing Operator shall have performed or complied in all material respects with each of its agreements and covenants required by this Agreement to be performed or complied with by it prior to or at the Effective Time.
- (h) Adverse Change. Except as otherwise provided in this Agreement, from the Effective Date until the Closing Date, there shall have been no material adverse change in the physical condition of the Assets (or any portion thereof), the condition of Existing Operator's licenses, permits and/or certificates, or the status of any Facility as a skilled nursing facility certified for participation in the Medicare and Medicaid programs, respectively.
 - (i) Liens. The Assets shall be free and clear of all Liens other than Permitted Liens.
- (j) Survey Reports. All Survey Reports with respect to the Facility shall have been resolved in accordance with Section 2.14(a).
- (k) Court Approval. Approval of this Agreement and the transactions contemplated thereby by the Court in any Bankruptcy Action.

- (l) Bond Escrow. New Operator shall have funded the Bond Escrow if required pursuant to Section 2.2(b).
- 1.8 <u>Conditions Precedent to Existing Operator's Obligation to Close.</u> The obligation of the Existing Operator to consummate the transactions described in this Agreement, and to take the actions required to be taken by Existing Operator, at the Closing and following the Effective Time is subject to the satisfaction, at or prior to Closing, of each of the following conditions (any of which may be waived by Existing Operator, in whole or in part):
- (a) Accuracy of Representations. Each representation and warranty of the New Operator in this Agreement must be accurate in all material respects as of the Closing Date as if made on the Closing Date, except for those representations and warranties which refer to facts existing at a specific date, which shall be accurate in all material respects as of such date.
- (b) Deliveries. At or prior to Closing, the Transferring Parties shall have received the deliveries under Section 1.6.
- (c) No Injunction. There must not be in effect any injunction, judgment, order, decree, ruling, or charge under any action, suit or proceeding before any court or quasi-judicial or administrative agency of any federal, state, local, or foreign jurisdiction or before any arbitrator that prohibits transfer of operations of, or the Assets used in connection with, the Facility.
- (d) Performance of Covenants. New Operator shall have performed or complied in all material respects with each of its agreements and covenants required by this Agreement to be performed or complied with by it prior to or at the Effective Time.
- (e) Court Approval. Approval of this Agreement and the transactions contemplated thereby by the Court in any Bankruptcy Action.

SECTION 2 TRANSFER OF OPERATIONS

2.1 <u>Cooperation</u>. The parties hereto agree to cooperate with each other to effect an orderly transfer of the operation of the Facility. Following the execution hereof, Existing Operator shall cooperate with New Operator to furnish all requested documentation in Existing Operator's possession and to execute all documents and consents reasonably necessary for New Operator to obtain any required licenses, agreements, certificates and consents necessary to operate the Facility not already in possession of New Operator, from third parties and government program agencies.

2.2 <u>Patient Funds; Advance Payments.</u>

- (a) On the Closing Date, and subject to adjustment within thirty (30) days following the Closing Date, Existing Operator shall provide New Operator with an accounting of all funds belonging to residents at the Facility, which are held by Existing Operator in a custodial capacity, and an accounting of all advance payments received by it pertaining to residents at the Facility (collectively, the "Funds/Payments"). Such accounting shall set forth the names of the residents for whom such Funds/Payments are held and the amounts held on behalf of each resident as of the Effective Time.
- (b) On the Closing Date, and subject to adjustment within thirty (30) days following the Effective Date, Existing Operator shall transfer the Funds/Payments to a bank account designated in

writing by New Operator. If New Operator is unable to establish a bank account for receipt of the Funds/Payments prior to Closing, Existing Operator shall permit New Operator to utilize the accounts of Existing Operator in which the Funds/Payments are deposited until such time as New Operator establishes its own account. In the event that any bond is required by the applicable banks or any Governmental Authority in order to allow New Operator to utilize Existing Operator's accounts in which the Funds/Payments are deposited, Existing Operator agrees that it shall post such bond with the applicable banks and/or Governmental Authorities, and New Operator shall escrow an amount at Closing equal to the amount of such bond (the "Bond Escrow") for the purpose of reimbursing Existing Operator for any Losses incurred as a result of such banks or Governmental Authorities making claims against Existing Operator with respect to such bond. The Bond Escrow shall be governed by an escrow agreement reasonably acceptable to New Operator and Existing Operator (the "Bond Escrow Agreement"). New Operator hereby acknowledges receipt of and expressly assumes all of Existing Operator's financial and custodial obligations arising subsequent to the Effective Time with respect to the Funds/Payments, it being the intent and purpose of this provision that, as of the Closing Date, Existing Operator is relieved of all financial and custodial obligations with respect to such funds and that New Operator hereby assumes all such post-Closing obligations and is directly accountable to the residents with respect thereto.

- (c) As of the Effective Time, New Operator shall assume custody of all trust accounts for residents of the Facility transferred by Existing Operator to New Operator and agree to treat such accounts in the fiduciary capacity required by law. New Operator agrees to indemnify and hold Existing Operator harmless from all liabilities, claims, and demands that may be asserted against Existing Operator in connection with New Operator's custody and treatment of such accounts from and after the Closing Date.
- 2.3 <u>Final Cost Reports</u>. Existing Operator shall prepare and file with the appropriate governmental authorities a final cost report for the Facility within the time frame required by law.

2.4 Employees of Existing Operator.

- (a) Existing Operator shall terminate employment and all benefits of the Facility' employees as of 11:59 p.m. MST on the day prior to the Closing Date. Existing Operator shall be responsible for and shall pay all wages due to all of the employees through 11:59 p.m. MST on the day prior to the Closing Date. New Operator shall offer to rehire at least 70% of the employees who work at the Facility for at least twenty (20) hours per week with benefits similar to what New Operator provides to its current employees in similar capacities (subject to Section 2.4(e)), provided, such employees meet New Operator's current hiring requirements. Those persons who receive and accept an offer of employment are referred to herein as the "Transitioned Employees". Notwithstanding the foregoing, New Operator shall not be obligated to hire any employee not approved by New Operator, including, without limitation, any employee who does not meet New Operator's current standards for employment.
- (b) New Operator and Existing Operator acknowledge and agree that the provisions of Section 2.4(a) are designed solely to ensure that Existing Operator is not required to give notice to the employees of the Facility of the "closure" thereof under the Worker Adjustment and Retraining Notification Act or under any comparable State law (collectively, the "WARN Act"). Nothing in Section 2.4 shall, however, create any rights in favor of any person not a party hereto, including the employees of the Facility, or constitute an employment agreement or condition of employment for any employee of Existing Operator or any affiliate of Existing Operator who is a Transitioned Employee. All Transitioned Employees shall be subject to the employment policies of New Operator after the Closing Date.
 - (c) To the extent that any obligations under the WARN Act arise with respect to the

employees at the Facility solely as a consequence of New Operator's failure to offer employment to a minimum number of employees of Existing Operator that results in an employment loss (as defined in the WARN Act) of fifty (50) or more employees within the applicable window under the WARN Act, New Operator shall be responsible for, and shall indemnify and hold harmless Existing Operator from and against any WARN liabilities, losses, or claims arising from New Operator's failure to continue to employ the requisite number of employees as required under the WARN Act.

- (d) Schedule 2.4(d) lists all earned and accrued vacation of all applicable Transitioned Employees ("ETO") as of the Closing Date. Existing Operator shall deliver to New Operator an updated Schedule 2.4(d) within five (5) days prior to the Closing which shall include all ETO (a) as of the most recent pay period preceding such Closing Date, and (b) as of such Closing Date. New Operator shall be responsible for ETO of all Transitioned Employees (the "Assumed ETO") after the Closing Date. Existing Operator shall be responsible for ETO of all employees other than Transitioned Employees. Subject to Section 2.4(e), Existing Operator shall be responsible for paying or providing all applicable employees with all other benefits to be provided to such employees for all periods prior to the Closing Date. Existing Operator shall provide a schedule of all ETO with respect to the Transitioned Employees to New Operator within five (5) Business Days after Closing. Existing Operator shall pay New Operator the aggregate amount of all Assumed ETO for the Transitioned Employees as of the Closing Date on the date of the next regular scheduled payroll date following the Closing.
- Existing Operator does not believe it has any obligation under applicable law to provide Transitioned Employees with health benefits after the Closing Date; nevertheless Existing Operator agrees to enroll each eligible Transitioned Employee and qualified beneficiary in "Continuation Coverage" as of the Closing Date (as that term is defined by Section 4980B of the Internal Revenue Code and Section 601, et seq. of ERISA ("COBRA")) for Existing Operator's Employee Plans that are medical, dental and vision plans for the period from the Closing Date until December 31, 2018 (the "COBRA Period"). Transitioned Employees shall only pay the current amount of premium payments that they are paying under the medical, dental and vision plans as of the date immediately prior to closing. New Operator shall reimburse Existing Operator and its Affiliates for: (i) the entire COBRA premium amount payable by the Transitioned Employee during the COBRA Period, (ii) any COBRA administrative fees payable by Existing Operator with respect to the COBRA Period, (iii) the amount of any claim made against any self-insured Employee Plans during the COBRA Period over the aggregate COBRA premium amount paid by all the Transitioned Employees on COBRA during the COBRA Period that are not covered by an existing stop loss insurance policy for the self-insured Employee Plans. The COBRA premium amount and administrative fees for November 2018 shall be paid at closing, and for December shall be paid on December 1, 2018. The parties shall determine the extent, if any that actual claim exceeded aggregate premiums paid one hundred and eighty (180) days from Closing, and any amount due Existing Operator shall be paid within ten (10) days thereafter. New Operator shall not reimburse Existing Operator for claims incurred under any self-insured Employees Plans in the period prior to the Closing Date. For the period following the COBRA Period, New Operator shall provide benefits to the Transitioned Employees similar to what New Operator provides to its current employees in similar capacities, and New Operator shall take no action which in any manner might encourage Transitioned Employees to continue to exercise their COBRA rights against Existing Operator following the COBRA Period. To the extent there is any liability for any "Continuation Coverage" for terminated employees of Existing Operator from and after the Closing Date, Existing Operator shall continue to be solely liable and responsible for, and New Operator shall incur no liability or responsibility with respect to, any Continuation Coverage for any employee or dependent of Existing Operator terminated at any time prior to the Closing Date.
- (f) Existing Operator acknowledges that no Employee Plans are considered part of the Assumed Liabilities, and any liabilities, claims or obligations with respect to such Employee Plans shall

remain with the Existing Operator.

2.5 Accounts Receivable.

- (a) New Operator shall assume responsibility for the billing for and collection of payments on account of services rendered by it at the Facility on and after the Closing Date. Existing Operator shall retain all rights in and title to billed and unbilled accounts receivable for services rendered at the Facility prior to the Closing Date (the "Existing Operator's Accounts Receivable") and shall retain full responsibility for the collection thereof. The Existing Operator's Accounts Receivable shall include all amounts due Existing Operator, whether billed or unbilled, prior to the Closing Date, for all services and ancillary services or products provided to any current or former residents at the Facility by Existing Operator prior to the Closing Date and any accounts receivable arising from the rate adjustments which relate to periods prior to the Closing Date even if such adjustments occur after the Closing Date.
- All third party payor payments that designate the dates of service or other identifying data on the remittance received by New Operator or Existing Operator from and after the Closing Date shall relate to the period prior to the Closing Date or after the Closing Date, as the case may be, in connection with the account of the resident of the Facility for whom the payment is made in accordance with the dates of service or other identifying data indicated on the remittance. New Operator shall remit to Existing Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date, provided such date is at least five (5) Business Days following receipt of such payment by New Operator, any third party payor payments received by New Operator that relate to the period prior to the Closing Date. Existing Operator shall remit to New Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date, provided such date is at least five (5) Business Days following receipt of such payment by Existing Operator, any third party payor payments received by Existing Operator that relates to the period on or after the Closing Date. In the event payment is made without remittance advice (or where the remittance advice does not specify the dates of service) and within sixty (60) days of the Closing Date, such payment will be first applied to the appropriate Existing Operator's Accounts Receivable, unless Existing Operator has no then outstanding accounts receivable with respect to the applicable resident or patient for whom such payment is made, in which case such payment will be applied to New Operator's accounts receivable. In the event payment is made without remittance advice (or where the remittance advice does not specify the dates of service) and subsequent to the sixtieth (60th) day following the Closing Date, such payment will be applied to the New Operator's accounts receivable, unless Existing Operator properly and fully identified such payment to New Operator before Closing, as updated within thirty (30) days following the Closing Date for residents' and patients' ancillary charges for the month immediately preceding the Closing Date. Notwithstanding anything contained herein to the contrary, advance payments received from private pay residents and patients billed by New Operator shall be retained by New Operator; provided such payments are for services received after the Effective Time. Any payments received by New Operator or Existing Operator during the first fifteen (15) days after the Closing Date from or on behalf of private pay patients with outstanding balances as of the Closing Date which fail to designate the period to which they relate, will first be applied to Existing Operator's Accounts Receivable, with any excess applied to reduce any balances due for services rendered by New Operator after the Closing Date. To the extent such payments are received by New Operator, New Operator shall remit any amounts to be applied to Existing Operator's Accounts Receivable to Existing Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date provided such date is at least five (5) Business Days following receipt of such payment by New Operator. To the extent such payments are received by Existing Operator, Existing Operator shall remit any amounts in excess of Existing Operator's Accounts Receivable to New Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date provided such date is at least five (5) Business

Days following receipt of such payment by Existing Operator. Following the fifteenth (15th) day after the Closing Date, all non-designated payments received by New Operator or Existing Operator from or on behalf of private pay patients with outstanding balances will first be applied to New Operator's accounts receivable, unless Existing Operator properly and fully identified such payment to New Operator before Closing, as updated within thirty (30) days following the Closing Date. To the extent such payments are received by New Operator, New Operator shall remit any amounts to be applied to Existing Operator's Accounts Receivable to Existing Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date provided such date is at least five (5) Business Days following receipt of such payment by New Operator. To the extent such payments are received by Existing Operator, Existing Operator shall remit any amounts to be applied to New Operator's accounts receivable to New Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date provided such date is at least five (5) Business Days following receipt of such payment by Existing Operator.

- (c) In addition, New Operator shall remit to Existing Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date, provided such date is at least five (5) Business Days following receipt of such payment by New Operator, any repayment or reimbursement received by New Operator arising out of cost reports filed for the cost reporting period ending prior to the Closing Date. Existing Operator agrees that any payment, whether received by New Operator from private pay residents and patients or as repayment or reimbursement arising out of cost reports, that pertain to services provided in the period commencing after the Effective Time shall be remitted by Existing Operator to New Operator on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date, provided such date is at least five (5) Business Days following receipt of such payment by Existing Operator, to be applied and/or disbursed by New Operator in accordance with the terms of this Section.
- (d) To the extent either party receives any proceeds from the accounts receivable of the other party, the parties acknowledge that the party receiving the payment belonging to the other party shall hold the payment in trust. Neither party shall have any right to offset with respect to such accounts receivable, and the party erroneously receiving the payment shall have no right, title or interest whatsoever in the payment and shall remit the same to the other on or before the next to occur of the tenth (10th) day or the twenty-fifth (25th) day of each calendar month after the Closing Date provided such date is at least five (5) Business Days following receipt of such payment by the applicable party. For a period of six (6) months following the Closing Date, Existing Operator and New Operator shall provide each other with an accounting by the twenty-fifth (25th) day of each month setting forth all amounts received by either of them during the preceding month with respect to accounts receivable. Nothing herein shall be deemed to limit in any way either party's rights and remedies to recover accounts receivable due and owing to it under the terms of this Agreement.
- (e) Existing Operator Accounts Receivable under Medicare or Medicaid are deposited into a control account (the "Government Receivables Account") at Wells Fargo Bank, National Association ("Wells Fargo") and swept daily by Wells Fargo and applied to Existing Operator's indebtedness under its line of credit with Wells Fargo.
- (f) The parties acknowledge that New Operator will use the Existing Operator's Medicare and/or Medicaid provider numbers for New Operator Accounts Receivable under Medicare and/or Medicaid (collectively, "Government Receivables"), and that the Government Receivables representing New Operator Accounts Receivable will be paid to the account currently designated for Existing Operator's Medicare and/or Medicaid provider numbers and swept daily by Wells Fargo Bank N.A. and applied to Existing Operators indebtedness due and owing to Wells Fargo Bank, N.A. until New Operator receives acknowledgement of the ownership change from Medicare and/or Medicaid, as the case may be

(the "tie-in notices"). On the Closing Date, Existing Operator shall provide New Operator with its NPI number and cooperate with Medicaid to provide access to the Medicaid portal to enable New Operator to submit billings and receive remittance advice for Medicaid receivables payable on or after the Closing Date and until New Operator receives the tie-in notices (the "Transition Period"). In addition, Existing Operator shall cooperate with New Operator to grant New Operator access to the Financial Intermediary Standard System in order to facilitate New Operator's billing of Medicare receivables with respect to the Transition Period.

- (g) The deposit of funds in the Government Receivables Account will not alter any person's right to or interest in or create a claim against such funds that does not otherwise exist. However, funds deposited in the Existing Operator's Government Receivables Account are swept by Wells Fargo as soon as practicable, and in any event, on or before Friday of each week and applied to Existing Operator's outstanding indebtedness under Wells Fargo line of credit. Existing Operator shall determine the allocation of all post-Closing Date collections pursuant to the terms and provisions of this Agreement and pay funds to New Operator consistent with this Agreement.
- (h) Deposits in the Government Receivables Account under Existing Operator's Medicare and/or Medicaid provider numbers will terminate once tie-in notices are issued by Medicare or Medicaid, as the case may be, such that New Operator Accounts Receivable from that government payor will thereafter be deposited directly into to an account owned only by New Operator. For each government payor, New Operator will promptly provide Existing Operator written notice when this condition is met.
- Payment of Operating Costs, Prorations and Deposits. Except as otherwise expressly provided herein, Existing Operator shall be responsible for, all claims or charges which are owed to third parties arising from Existing Operator's ordinary use, operation or control of the Facility, including payroll, taxes, insurance premiums, utilities, amounts due under executory obligations, amounts due to any third party vendors, including amounts due under any Assumed Contracts, and similar obligations for all periods prior to the Closing Date. New Operator, to the extent it utilizes the services provided by third parties, shall be responsible for, and shall pay on a timely basis, any claims or charges which are due to such third parties arising from the use, operation or control of the Facility from and after the Effective Time. Revenues and expenses pertaining to utility charges for the billing period in which this Agreement is executed, prepaid expenses and like items of revenue or expense attributable to the Facility shall be prorated between Existing Operator and New Operator as of the Effective Time. All such prorations shall be made on the basis of actual days elapsed in the relevant accounting or revenue period and shall be based on the most recent information available to Existing Operator. Utility charges that are not metered and read on the Closing Date shall be estimated based on prior charges, and shall be re-prorated upon receipt of statements therefor. In general, such prorations shall be made so as to reimburse Existing Operator for actual prepaid expense items, and to charge Existing Operator for prepaid revenue items, to the extent that the same are attributable to periods after the Effective Time. To the extent any third party utility provider delivers final readings as of the Closing Date, the Existing Operator shall be responsible for paying such final invoice amount.
- 2.7 <u>Treatment of Prorations</u>. The accounts of Existing Operator and New Operator created pursuant to the prorations provided for in the preceding Section 2.4 and Section 2.6 shall be netted against each other and paid to Existing Operator or New Operator, as applicable, on the Closing Date.
- 2.8 <u>Future Settlement</u>. All amounts owing from one party hereto to the other party hereto, excluding amounts in respect of Section 2.2 hereof, that require adjustment after the Closing Date, including without limitation, re-prorations according to Section 2.6 hereof, shall be settled three (3) months after the Closing Date. If, thereafter, a party hereto determines that any further adjustment is to be made, such party shall submit a statement to the owing party setting forth any and all such items and the calculation of the

amounts due hereunder. Such statement shall be submitted with appropriate backup materials. If amounts are owing from New Operator to Existing Operator, New Operator shall have thirty (30) days from the date of receipt of such statement to tender payment to Existing Operator or to question or dispute in writing any item thereon. If amounts are determined to be owing from Existing Operator to New Operator, Existing Operator shall have thirty (30) days from the date of receipt of such statement to tender payment to New Operator or to question or dispute in writing any item thereon.

2.9 Medicare/Medicaid

- On the Closing Date, Existing Operator shall assign to New Operator its Medicare and Medicaid provider numbers and agreements in accordance with and as permitted by any and all applicable laws and orders, rules, requirements and regulations of the United States Department of Health and Human Services' Centers for Medicare & Medicaid Services ("CMS") and the state where the Facility is located subject to any and all other applicable federal or state statutes and regulations regarding the same. Existing Operator and New Operator acknowledge and agree that New Operator is not expected to have received "tie in" notice from CMS with respect to Existing Operator's Medicare or Medicaid provider agreement or a new Medicare or Medicaid provider agreement as of the Effective Time. Accordingly, prior to New Operator's receipt thereof, Existing Operator agrees that New Operator may begin to bill for its services under Existing Operator's Medicare and Medicaid provider numbers, as applicable, to the extent permitted as aforesaid, pending receipt of the tie in notices/new Medicare and Medicaid provider agreements, and the required provider numbers have been received by New Operator; provided, that, New Operator shall indemnify and hold Existing Operator and its Affiliates, directors, officers, limited partners and general partner harmless from any and all liabilities and claims related to New Operator's use of Existing Operator's Medicare and Medicaid provided numbers. Nothing set forth herein shall be deemed to limit in any way Existing Operator's right, title, and interest in its cash and accounts receivable for services rendered prior to the Closing Date, which cash and accounts receivable are property of the Existing Operator and shall be reimbursed or retained, as applicable, in accordance herewith.
- (b) Existing Operator and New Operator understand that reimbursements from Medicare and/or Medicaid for items/services provided/rendered after the Closing Date may continue to be issued to Existing Operator for a period of time. If and to the extent that they include payment for periods on and after the Closing Date and New Operator is entitled to such payments in accordance with the priority for such remittances as set forth in Section 2.5(b) of this Agreement, Existing Operator shall forward all such Medicare and/or Medicaid reimbursements in accordance with Section 2.5 until such time as reimbursements are remitted directly to New Operator by Medicare and Medicaid.
- <u>Transfer of Records</u>; <u>Access To Records</u>. As of the Closing Date, Existing Operator shall transfer 2.10 to New Operator the records of all Transitioned Employees and all then current residents in the Facility (the "Transferred Records") by leaving all such records at the Facility, provided, however, that Existing Operator shall be entitled to keep such copies of all Transferred Records as it may deem necessary and as permitted by law. For the avoidance of doubt, the Transferred Records shall include any Occupational Health and Safety ("OSHA") records required to be maintained in accordance with OSHA recordkeeping regulations (29 CFR 1904), including, without limitation, any OSHA Form 300, Form 300A and Form 301 records. From and after the Closing Date, New Operator shall be solely responsible for caring for the residents of the Facility in accordance with their contractual rights and in accordance with law. New Operator shall preserve the existence and maintain the confidentiality of the resident records transferred to New Operator pursuant to this Agreement in accordance with federal and state law including HIPAA. Subsequent to the Closing Date, New Operator shall allow Existing Operator and its affiliates, agents and representatives, at Existing Operator's sole cost and expense, to have reasonable access during regular business hours upon reasonable prior written notice and to make copies of, the Transferred Records, to the extent reasonably necessary to enable Existing Operator to investigate and defend malpractice, employee

or other claims, to file or defend cost reports and tax returns, to verify accounts receivable collections due Existing Operator, and to perform similar matters. New Operator will maintain the Transferred Records, to the extent required by law, but in no event less than seven (7) years from the Closing Date with respect to resident records, and no less than six (6) years from the Closing Date with respect to other records

- 2.11 <u>Deposits</u>. All deposits, if any, held by a utility or other party to an executory contract shall remain the property of Existing Operator, and New Operator shall be required to post its own replacement security deposits, including, but not limited to, any security deposits required under any Assumed Contracts. Any such deposits remaining the property of Existing Operator as contemplated herein that are received by New Operator shall be forwarded to Existing Operator within 15 Business Days of New Operator's receipt of any such deposit.
- 2.12 <u>Compliance with Laws</u>. The parties shall comply in all material respects with all applicable laws, and with all applicable rules and regulations of all governmental authorities, in conjunction with the execution, delivery and performance of this Agreement and the transactions contemplated hereby.
- 2.13 <u>Accounts Payable</u>. New Operator agrees to assume all accounts payable for supplies and other goods or equipment ordered in the ordinary course of business and received at the Facility on and subsequent to the Effective Time, and for services rendered and performed on and subsequent to the Effective Time; provided, however, that New Operator does not assume and shall not be responsible for any accounts payable for any goods or services delivered or performed at the Facility pursuant to any contract that is not as Assumed Contract.

2.14 <u>Regulatory Inspections; Surveys.</u>

- With respect to any Survey Report issued prior to the Closing Date, Existing Operator agrees that any citations and/or deficiencies are the responsibility of, and must be corrected and/or paid for, by Existing Operator. Existing Operator's responsibility will include correcting all non-compliances and/or citations, paying any and all fines, providing a plan of correction, providing and bearing the expense for all consultants, staff, materials, supplies and equipment necessary to complete the plan of correction and achieve full compliance. In the event there is a cost incurred to resolve any conditions relating to the Facility or the Assets subject to and required by Governmental Authorities as a result of a Survey Report issued with respect to the Facility between the Effective Date and the Closing Date, including, without limitation, a survey and/or re-licensing inspection by a Governmental Authority arising from the transaction contemplated in this Agreement, Existing Operator shall have five (5) Business Days from the date such cost is determined to advise New Operator whether or not it is willing to pay such cost. The failure of Existing Operator to respond within said five (5) Business Day period shall be deemed to be an election by Existing Operator not to pay. If Existing Operator determines to pay less than all of the costs, New Operator shall have five (5) Business Days thereafter in which to advise Existing Operator in writing of New Operator's election (x) to pay any additional costs and proceed to Closing or (y) to terminate this Agreement in accordance with Section 8. The failure of New Operator to respond within said five (5) Business Day period shall be deemed to be an election by New Operator to terminate this Agreement. Existing Operator shall have the right to control the manner in which such conditions are resolved. New Operator shall not do anything to encourage a finding that there are any such conditions that must be resolved. In the event of a survey or investigation of the Facility by any Health Care Authority, Existing Operator and New Operator agree that the time for Closing shall be extended, if necessary, to receive the Survey Report related to such survey or investigation and to resolve, in accordance with this Section 2.14(a), any conditions relating to such Facility or the Assets subject to and required by such Health Care Authority as a result of such Survey Report.
 - (b) In connection with survey and re-licensing matters, Existing Operator and New Operator

agree to cooperate fully with each other in preparing, filing, prosecuting, and taking any other actions with respect to any applications, requests, or actions that are or may be reasonable and necessary to obtain the consent of any governmental instrumentality.

- (c) From and after the Closing Date, Existing Operator shall be responsible for the payment of all fines and penalties imposed by Governmental Authorities which fines and penalties arise in connection with any regulatory investigations, inspections or surveys occurring prior to the Effective Time or which are related to the operations of the Facility prior to the Effective Time. New Operator shall be responsible for the payment of all fines and penalties imposed by Governmental Authorities which fines and penalties arise in connection with any regulatory inspections or surveys occurring on or after the Effective Time and which are related to the operations of the Facility on or after the Effective Time. Existing Operator shall assist, as appropriate, New Operator in establishing any plans of correction or other responses to be submitted by New Operator after the Effective Time for surveys or inspections that relate to events or circumstances occurring both prior to and on or after the Effective Time within the time allowed for such submissions; however, New Operator's determination as to such plan of correction or other submission shall be controlling and New Operator shall be responsible for and bear all costs and expenses (including fines for any delays in submission) as a result of implementing such plan of correction or other submission.
- 2.15 <u>Delivery of Inventories and Supplies</u>. At Closing, Existing Operator shall deliver to New Operator by leaving at the Facility on the Closing Date such inventory and supplies in condition, quantity and quality sufficient to meet the regular and customary operating needs for operating in accordance with law and the previously established policies of Existing Operator at the Facility. Any business records being acquired by New Operator pursuant to this Agreement shall be delivered by leaving all such records at the Facility except to the extent electronic copies of such records have been provided to New Operator prior to the Closing Date.
- 2.16 <u>Assumed Contracts</u>. At Closing, Existing Operator shall assign to New Operator and New Operator shall accept assignment of (i) all Resident Contracts and (ii) all Assumed Contracts. New Operator shall be obligated to pay all amounts arising under any Resident Contracts or Assumed Contracts pursuant to Section 2.6 hereof until such time as such Resident Contracts or Assumed Contracts are terminated. Payments under the Resident Contracts or Assumed Contracts shall be prorated through the Closing Date in accordance with the terms of Section 2.6 above.
- Remittances, Mail and Other Communications. All remittances, mail and other communications relating to the operations of the Facility or the Resident Contracts or Assumed Contracts following the Closing Date received by Existing Operator or its affiliates at any time after the Closing Date relating to periods of operations on and after the Closing Date shall be promptly turned over to New Operator. All remittances, mail and other communications relating to the operations of the Facility or the Resident Contracts or Assumed Contracts prior to the Closing Date received by New Operator or its affiliates at any time after the Closing Date related to periods of operations prior to the Closing Date shall be promptly turned over to Existing Operator.
- 2.18 <u>Obligations of New Operator</u>. Any obligation this Agreement imposed upon New Operator shall not preclude New Operator from also imposing that obligation upon, or implementing that obligation through a management company; <u>provided</u>, <u>however</u>, that it is expressly understood and agreed that nothing contained in this sentence shall relieve New Operator from its liability for such obligations under this Agreement.
- 2.19 <u>Excluded Liabilities</u>. Except as specifically provided for in this Agreement, New Operator shall not assume any liabilities or obligations of Existing Operator whatsoever, fixed or contingent.

2.20 <u>IT Transition</u>.

- (a) For a period of ten (10) days following the Closing Date, New Operator shall have the right to use Existing Operator's data circuit operating system at the Facility. In addition, for a period of ninety (90) days following the Closing Date, New Operator shall have the right to use Existing Operator's voice circuit operating system and cable or satellite television contracts with respect to the Facility. Such operating systems and contracts shall continue following the Closing Date in a manner and at levels consistent with how such services were provided to the Facility immediately prior to the Closing Date. New Operator shall pay Existing Operator the "prevailing rate" (as defined herein) for the third party services provided to New Operator with respect to such operating systems and contracts (based on metered usage, where applicable). For purposes hereof, the term "prevailing rate" means the then current rate charged to Existing Operator by the provider of such services or utilities, including any applicable taxes and other charges.
- Existing Operator and New Operator agree that Existing Operator's computer hardware (b) that is included in the Excluded Assets may nonetheless remain at the Facility following the Closing Date, and Existing Operator shall remove such computer hardware from the Facility within twenty (20) days following the Closing Date. During such period, New Operator shall store such computer hardware in a secure location at the Facility and provide access to Existing Operator for the purposes of removing such equipment, which access shall be during regular business hours and granted in such a manner so as to not unreasonably interfere with New Operator's operation of the Facility. In the event that Existing Operator has not removed such computer hardware within twenty (20) days following the Closing Date. New Operator shall may dispose of such computer hardware in New Operator's discretion upon not less than five (5) business days' prior notice to Existing Operator. Any proceeds received by New Operator with respect to the disposition of such computer hardware shall be held for the account of Existing Operator; provided that, New Operator shall be permitted to deduct any out of pocket costs incurred by New Operator with respect to any such disposition from such proceeds. Notwithstanding anything herein to the contrary, Existing Operator hereby acknowledges and agrees that none of the New Operator, its affiliates, employees or consultants shall be liable for any damages or losses with respect to Existing Operator's computer hardware remaining at the Facility following the Closing Date, other than such damages or losses resulting from or arising out of the gross negligence or willful misconduct of the New Operator, its affiliates, employees or consultants.
- (c) For a period of thirty (30) days following the Closing Date, Existing Operator shall continue to operate, and shall provide New Operator with access to, its American HealthTech software applications and related data. In addition, for a period of thirty (30) days following the Closing Date, Existing Operator shall provide New Operator with access to its Resident Fund Management Service and American HealthTech resident trust fund data, as applicable.
- Managed Care Providers. To the extent permitted by such contracts and applicable law, Existing Operator agrees that New Operator may elect, in its sole discretion, to assume Existing Operator's managed care provider plans; provided that the applicable managed care providers acknowledge and agree that Existing Operator is released from its obligations under such managed care provider plans for all periods from and after that Closing Date. In the event that New Operator elects not the assume Existing Operator's managed care provider plans, to the extent permitted by such contracts and applicable law, Existing Operator agrees that New Operator may bill for its services under Existing Operator's managed care provider plans using Existing Operator's provider information from and after the Closing Date until such time as New Operator has obtained its own managed care provider plans. New Operator shall indemnify Existing Operator and its affiliates, directors, officers, limited partners and general partners for all liabilities in connection with its use of Existing Operator's managed care provider plans.

Non-Solicitation of Employees. From the Effective Date through the Closing Date and for a period of twelve (12) months after the Effective Time, Existing Operator shall not, nor shall Existing Operator assist others to, solicit or induce employees of the Facility (whether employed by such Facility as of the Closing Date or at some future date) to alter or terminate their relationships with such Facility (or the New Operator, post-Closing); provided, that, with respect to employees of the Facility immediately prior to the Closing Date (and not to future employees of such Facility after the Closing Date), the foregoing restriction shall not apply to non-Transitioned Employees. Nothing herein shall be construed to extend to the employees of the current management company, including regional and divisional employees.

SECTION 3 REPRESENTATIONS AND WARRANTIES OF NEW OPERATOR

New Operator hereby makes the representations and warranties indicated below to Existing Operator:

- Authority, Validity and Binding Effect. New Operator has all necessary limited liability company power and authority to operate and lease the Facility and to carry on its business as it is now being conducted. New Operator is duly organized and in good standing under the laws of its jurisdiction of domestic organization. New Operator has all necessary limited liability company power and authority, as the case may be, to enter into this Agreement and to execute all documents and instruments referred to herein or contemplated hereby and all necessary action has been taken to authorize the individuals executing this Agreement on each of their behalf to do so. This Agreement has been duly validly executed and delivered by New Operator and is enforceable against New Operator in accordance with its terms, except as limited by bankruptcy, insolvency, fraudulent conveyance, moratorium, liquidation, reorganization or other similar laws affecting the enforcement of creditors' rights in general.
- 3.2 <u>No Defaults</u>. The execution and delivery of this Agreement and any documents contemplated hereby by New Operator, and the performance of its obligations hereunder and thereunder, do not and will not:
- (a) conflict with or result in any material breach of the provisions of, or constitute a default under, the certificate of formation or operating agreement of New Operator;
- (b) violate any material restriction to which New Operator is subject or, without the giving of notice, passage of time, or both, violate (or give rise to any right of termination, cancellation or acceleration under) any material license, authorization or permit or other material agreement or instrument to which New Operator is a party, which will not be satisfied or terminated prior to the Closing Date as a result of the transactions contemplated by this Agreement or result in the termination of any such instrument or termination of any provisions in such instruments that will result in the impairment of any of New Operator's rights under such instruments; and
- (c) constitute a violation of any applicable material resolution, rule, regulation, law, statute or ordinance of any administrative agency or governmental authority, or of any judgment, decree, writ, injunction or order of any court to which New Operator is subject or by which its assets are bound, or any credit agreement or other financing arrangement to which New Operator, or any of its affiliates is a party.
- 3.3 <u>Governmental Authorities</u>. Except as disclosed in <u>Schedule 3.3</u>, New Operator is not required to submit any notice, report or other filing with any federal, state, municipal, foreign or other governmental or regulatory authority in connection with its execution or delivery of this Agreement or any of the New

Operator Documents or the consummation of the transactions contemplated hereby and no consent, approval or authorization of any governmental or regulatory authority is required to be obtained by New Operator in connection with the execution, delivery and performance of this Agreement or the New Operator Documents or the consummation of the transactions contemplated hereby or thereby. To New Operator's Knowledge, there are no circumstances that exist that would prevent New Operator from obtaining the consents and approvals set forth on Schedule 3.3 in the ordinary course of business.

3.4 <u>HIPAA</u>. New Operator is a "Covered Entity" as defined under Health Insurance Portability and Accountability Act of 1996, as amended by the Health Information Technology for Economic and Clinical Health Act ("HIPAA"). The transactions contemplated by this Agreement will not violate any applicable data privacy or security requirements under HIPAA or any privacy or security requirements (other than HIPAA Commitments) imposed by federal or state law on healthcare data, including state healthcare data breach notification laws and related state consumer protection laws.

3.5 Intentionally Omitted.

3.6 <u>Litigation</u>. There are no claims, actions, suits, proceedings or investigations pending or to New Operator's Knowledge threatened by or against New Operator with respect to this Agreement, or in connection with the transactions contemplated hereby, and to New Operator's Knowledge, there is no other reason to believe there is a valid basis for any such claim, action, suit, proceeding or investigation.

SECTION 4 REPRESENTATIONS AND WARRANTIES OF EXISTING OPERATOR

Existing Operator hereby represents and warrants as follows to New Operator:

- Authority, Validity and Binding Effect. Existing Operator has all necessary corporate power and authority to carry on its business as it is now being conducted. Existing Operator is duly organized under the laws of the State of Texas. Existing Operator has all necessary corporate power and authority to enter into this Agreement and to execute all documents and instruments referred to herein or contemplated hereby and all necessary action has been taken to authorize the individuals executing this Agreement to do so subject to Court approval. This Agreement has been duly and validly executed and delivered by Existing Operator and upon Court approval shall be enforceable against Existing Operator in accordance with its terms, except as limited by bankruptcy, insolvency, fraudulent conveyance, moratorium, liquidation, reorganization or other similar laws affecting the enforcement of creditors' rights in general.
- 4.2 <u>No Defaults</u>. The execution and delivery of this Agreement and any documents contemplated hereby by Existing Operator, and the performance of their obligations hereunder, does not and will not:
- (a) conflict with or result in any material breach of the provisions of, or constitute a default under the certificate of formation or operating agreement of Existing Operator;
- (b) violate any material license, authorization or permit or other material agreement or instrument to which Existing Operator is a party, which will not be satisfied or terminated with respect to the Facility prior to the Closing Date as a result of the transactions contemplated by this Agreement or result in the termination of any such instrument or termination of any provisions in such instruments that will result in the impairment of any of Existing Operator's rights under such instruments; or
- (c) subject to approvals in the Bankruptcy Action, constitute a violation of any applicable material resolution, rule, regulation, law, statute or ordinance of any administrative agency or governmental authority, or any judgment, decree, writ, injunction or order of any court to which Existing

Operator is subject or by which its assets are bound, or any credit agreement or other financing arrangement to which Existing Operator is a party.

4.3 <u>HIPAA</u>. Existing Operator is a "Covered Entity" as defined under HIPAA. The transactions contemplated by this Agreement will not violate any applicable data privacy or security requirements under HIPAA or any privacy or security requirements (other than HIPAA Commitments) imposed by federal or state law on healthcare data, including state healthcare data breach notification laws and related state consumer protection laws.

4.4 Health Care Matters.

- (a) All material Medicare and Medicaid provider agreements, certificates of need, if applicable, certifications, governmental licenses, permits, regulatory agreements or other agreements and approvals, including certificates of operation, completion and occupancy, and state nursing facility licenses or other licenses required by Health Care Authorities (as defined below) for the legal use, occupancy and operation of the Facility (collectively, "Health Care Licenses") have been obtained by the party required to hold such Health Care Licenses and are in full force and effect, including approved provider status in any approved third-party payor program. Existing Operator owns and/or possesses, and holds free from restrictions or conflicts with the rights of others, all such Health Care Licenses.
- (b) The Facility is duly licensed as a skilled nursing facility as required under the applicable laws of the state in which the Facility is located. Existing Operator has not applied to reduce the number of licensed or certified beds of the Facility or to move or transfer the right to any and all of the licensed or certified beds of the Facility to any other location or to amend or otherwise change the Facility and/or the number of beds approved by the state health department or equivalent (or any subdivision) or other applicable state licensing agency, and there are no proceedings or actions pending or contemplated to reduce the number of licensed or certified beds of the Facility.
- (c) The Health Care Licenses (i) have not been (A) transferred to any other location or (B) pledged as collateral security unless such pledge will be released at Closing; and (ii) are not provisional, probationary, or restricted in any way.
- (d) Existing Operator has not taken any action to rescind, withdraw, revoke, amend, modify, supplement or otherwise alter the nature, tenor or scope of any Health Care License or applicable provider payment program other than non-material alterations effected in the ordinary course of business.
- (e) Existing Operator (and the operation of the Facility) is in substantial compliance with the applicable provisions of the laws, ordinances, statutes, regulations, requirements of participation, orders, standards, policies, restrictions or rules of any Health Care Authority having jurisdiction over the Facility, other than matters disclosed on Schedule 4.4(e).
- (f) To the Existing Operator's Knowledge, Existing Operator is in material compliance with the terms of its provider agreements for Medicare and Medicaid and any other applicable requirements for participation in the Medicare and Medicaid programs.
- (g) Other than the currently pending litigation in New Mexico filed by the New Mexico Attorney General v. Preferred Care, Inc. et. al, state of New Mexico, County of Santa Fe, First Judicial District, D-101-CV-2014-02535, Existing Operator is not, to Existing Operator's Knowledge, a target of, and Existing Operator is not a participant in, or subject to any action, proceeding, suit, audit, investigation or sanction by any Health Care Authority or any other administrative or investigative body or entity or any other third party payor or any resident (including, without limitation, whistleblower suits, or suits

brought pursuant to federal or state False Claims Acts, and Medicare/Medicaid/State fraud and abuse laws, but excluding medical malpractice claims and other civil liability lawsuits which would reasonably be expected to result, directly or indirectly or with the passage of time, in the imposition of a material fine, penalty, alternative, interim or final sanction, a lower rate certification, recoupment, recovery, suspension or discontinuance of all or part of reimbursement from any Health Care Authority, third-party payor, insurance carrier or private payor, a lower reimbursement rate for services rendered to eligible residents, or any other civil or criminal remedy, or which could reasonably be expected to have a material adverse effect on Existing Operator or the operation of any Facility, nor has any such action, proceeding, suit, investigation or audit been threatened.

- (h) To the Existing Operator's knowledge, there are no written agreements with residents of the Facility or with any other persons or organizations that deviate in any material adverse respect from or that conflict with any statutory or regulatory requirements.
- (i) Existing Operator has instituted, and the Facility is operated in material compliance with, a compliance plan which follows applicable guidelines established by Health Care Authorities.
- (j) Existing Operator is in material compliance with the Health Care Insurance Portability and Accountability Act of 1996 and the Health Information Technology for Economic and Clinical Health Act, as incorporated in the American Recovery and Reinvestment Act of 2009, and the regulations promulgated under each.
- (k) There is no pending or, to Existing Operator's Knowledge, threatened revocation, suspension, termination, probation, restriction, limitation or non-renewal affecting Existing Operator, the Facility or any provider agreement relating to the Facility with any third-party payor or Medicare or Medicaid.
- (l) Existing Operator has delivered to New Operator true, correct and complete Resident Census Information for each Facility's last three (3) fiscal years and current year-to-date broken out by month.
- (m) Except as disclosed on Schedule 4.4(m), neither Existing Operator nor the Facility has any contract or agreement with, or is currently providing services to any patient whose payor source is, the Department of Veterans Affairs or any division thereof for the provision of services to patients or residents at any Facility. [NOTE: in the Arizona Sun City facility OTA, please disclose the applicable VA contracts.]
- 4.5 <u>Governmental Authorities</u>. Except as disclosed in <u>Schedule 4.5</u>, Existing Operator is not required to submit any notice, report or other filing with any federal, state, municipal, foreign or other governmental or regulatory authority in connection with its execution or delivery of this Agreement or any of the Existing Operator Documents or the consummation of the transactions contemplated hereby and no consent, approval or authorization of any governmental or regulatory authority is required to be obtained by Existing Operator in connection with the execution, delivery and performance of this Agreement or the Existing Operator Documents or the consummation of the transactions contemplated hereby or thereby.
- 4.6 <u>Taxes</u>. To Existing Operator's Knowledge, all Taxes have been paid in full to the extent due and payable, all Tax Returns required to be filed in connection therewith have been filed, and Existing Operator is not delinquent in the payment of any Tax, assessment or governmental charge in connection with the operation of any Facility and as of the Closing Date, will have no Tax deficiency or claim outstanding or assessed against it in connection with the operation of any Facility.

4.7 <u>Resident Admission Agreements</u>. To Existing Operator's Knowledge, none of the resident admission agreements (collectively, the "Resident Contracts") (i) entitles any resident to care at no cost, or (iii) entitles any resident to care for a reduced cost from that charged to other residents of the Facility.

4.8 Title Related Matters.

- (a) No labor has been performed or material furnished for the Facility for or on behalf of Existing Operator, in any material amounts, for which Existing Operator has not heretofore fully paid, or for which any mechanics' or materialmans' lien or liens, or any other lien, can be lawfully claimed by any person, party or entity.
- (b) There are no parties (other than Existing Operator) in possession of the Facility or the Assets, or any portion thereof, other than Existing Operator.

4.9 Environmental Matters.

- (a) To Existing Operator's Knowledge, No Facility contains any Hazardous Substance, except for Hazardous Substances typically used in, and in quantities necessary for the day-to-day operation of, any Facility and which are commonly used in other similar facilities, including but not limited to cleaning fluids, insecticides and medicines and which are not in violation of any Environmental Laws ("Common Products").
- (b) Existing Operator has received no written notice of any violation of applicable laws, nor is there any pending, or to Existing Operator's Knowledge, threatened, litigation or proceeding before any court of any governmental or administrative agency, concerning the violation of any Environmental Law.
- (c) Existing Operator has operated the Facilities in material compliance with all Environmental Laws.

4.10 Intentionally Omitted.

- 4.11 <u>Assets.</u> All of the tangible Assets are in Existing Operator's possession or control and located at or on the Facility. The FF&E which are material to the operation of the Facilities are in good repair and working order, subject to normal wear and tear, and adequate and suitable for the purposes for which they are presently being used.
- 4.12 <u>Survey Reports</u>. True and complete copies of all Survey Reports in Existing Operator's possession obtained in the previous three (3) years have been provided to New Operator.
- 4.13 <u>Capital Expenditures</u>. Except as disclosed in <u>Schedule 4.13</u>, as of the Effective Date, Existing Operator does not have any outstanding contracts for capital expenditures relating to the Facility, nor does it have any agreement, obligations or commitments for future capital expenditures relating to the Facility, including, without limitation, additions to property, plant, equipment or intangible capital assets.
- 4.14 <u>Absence of Notices</u>. Existing Operator has not received any written notice that (a) any material provider or supplier of Existing Operator intends to discontinue, substantially alter prices or

terms to, or significantly diminish its relationship with the Facility as a result of the transaction contemplated hereby or otherwise or (b) any of the managerial employees intends to terminate his or her employment at any Facility on account of the transactions contemplated hereby or otherwise.

- 4.15 <u>Resident Records.</u> All resident records and other relevant records of the Facility shall be retained and maintained by Existing Operator at the Facility through the Closing Date.
- 4.16 <u>Insurance</u>. Except as set forth on <u>Schedule 4.16</u>, all insurance coverage for fire, liability, worker's compensation, title and other forms of insurance owned or held by Existing Operator and applicable to the Assets or the business at the Facility are in full force and effect, and insurance certificates evidencing the same have been provided to New Operator.
- 4.17 <u>Litigation</u>. Except as disclosed in <u>Schedule 4.17</u>, there are no actions, suits, arbitrations, regulatory proceedings or other litigation, proceedings or governmental investigations pending or, to Existing Operator's Knowledge, threatened against or affecting Existing Operator or any of its officers, directors, employees or members related to the Facility.
- 4.18 <u>Intentionally Omitted.</u>
- 4.19 <u>Intentionally Omitted.</u>
- 4.20 <u>Compliance with Laws</u>. Existing Operator, the Assets and the operation of the Facility comply in all material respects with all Laws applicable to the Assets and operations conducted at the Facility.

4.21 <u>Employees</u>.

- (a) Schedule 4.21(a) is a true and complete list of all employees of Existing Operator. There are no unresolved labor controversies (including unresolved grievances and age or other discrimination claims), if any, between Existing Operator and any person employed by or providing services to Existing Operator which have not been disclosed to New Operator. Existing Operator is not a party to (i) any material employment agreement or similar arrangement, other than written agreements or arrangements that may be terminated at any time upon no more than ninety (90) days' prior notice without penalty or (ii) any material employment agreement that causes an employee to be other than an "at will" employee.
- (b) Existing Operator is not (i) a party to, involved in, subject to an order by a Governmental Authority arising out of or, to Existing Operator's Knowledge, threatened by, any labor or employee dispute or unfair labor practice charge or equal employment complaint or (ii) currently a party to or negotiating any collective bargaining agreement, and Existing Operator has not experienced any work stoppage by reason of employee action during the last three (3) years.
- Employee Benefit Plans. Except as disclosed in Schedule 4.22, neither Existing Operator nor any of its ERISA Affiliates is a party to, participates in, has participated in, has any employees participating in, or has any liability or contingent liability with respect to an Employee Plan. "Employee Plans" means (i) any "employee welfare benefit plan" or "employee pension benefit plan" or "multiemployer plan" as those terms are respectively defined in sections 3(1), 3(2) and 3(37) of the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder ("ERISA"); or (ii) any retirement or deferred compensation plan, incentive compensation plan, stock plan, unemployment compensation plan, vacation pay, severance pay, bonus or

benefit arrangement, insurance or hospitalization program or any other fringe benefit arrangements for any current or former employee, director, consultant or agent, whether pursuant to contract, arrangement, custom or informal understanding, written or unwritten, which does not constitute an "employee benefit plan" (as defined in section 3(3) of ERISA).

4.23 <u>Disclosure</u>. No representation or warranty by Existing Operator in this Agreement and no statement contained in the schedules attached hereto or any certificate or other document furnished or to be furnished to New Operator pursuant to this Agreement contains or will contain, as of the effective date of such disclosure, any untrue statement of a material fact or omits or will omit to state a material fact necessary to make the statements contained herein or therein, as of the effective date of such disclosure, in light of the circumstances in which they are made, not misleading.

SECTION 5 COVENANTS OF EXISTING OPERATOR

From the Effective Date and until the Closing and to the extent thereafter as contemplated herein, except as otherwise consented to or approved by New Operator in writing, Existing Operator covenants and agrees as set forth below. Where a provision includes an undertaking by New Operator, Existing Operator covenants and agrees as well.

- Regular Course of Business. Existing Operator shall (a) operate the Facility in a manner substantially consistent with Health Care Requirements, applicable Law and past practices; (b) use commercially reasonable efforts to comply in all material respects with the provisions of all Laws applicable to the operation of the Facility, including, without limitation, compliance with requirements of all government programs; (c) use its commercially reasonable efforts to maintain (except for expiration due to lapse of time), all Resident Contracts, leases and contracts in effect without change except as expressly provided herein or in the ordinary course of business consistent with past practices; (d) not enter into, any agreements or leases which would be binding upon New Operator (d) keep in full force and effect present insurance policies through the Closing Date or other comparable insurance coverage; and (e) use its commercially reasonable efforts to maintain in good standing all Health Care Licenses and other licenses necessary to operate the Facility.
- Absence of Certain Changes or Events. Except for increases and changes made in the ordinary course of business or pursuant to written contractual obligations in effect and in existence as of the Effective Date and disclosed to New Operator on Schedule 5.2 hereof, as to the employees of the Facility, and any agents and Affiliates of Existing Operator, there will not be any of the following: any new bonus, percentage compensation, service award or other like benefit, or any increase in the compensation payable or to become payable by Existing Operator to any of such employees in the ordinary course of business; or any change in the method of calculating any presently existing bonus, percentage compensation, service award or other like benefit, granted, made or accrued to or to the credit of any of such employees or agents of Existing Operator.

5.3 Full Access and Disclosure.

(a) Except as prohibited by law or as provided in this Agreement, upon reasonable notice and subject to reasonable conditions such as the condition that a representative of Existing Operator accompany New Operator, Existing Operator shall afford to New Operator, and its counsel, accountants and other authorized representatives reasonable access during business hours to the Assets, the Facility, books and records, in any way relating to the Assets and/or the Facility, including, but not limited to, the roofs, all equipment (fixed and movable), heating and cooling systems, and any and all vehicles, financial

data and records, operating data and other information reasonably requested, including the most recent financial statements, cost reports, inspection reports, plans of correction current room rates (including dates and amounts of increases), census data and resident mix, payroll information, Medicaid reports, employment agreements, personnel policies, occupancy agreements with residents, leases, and all contracts, agreements, and other documents relating to outside contractors, vendors, consultants, or other outside parties relating to the Facility and to which the Facility or Existing Operator is now or may become a party in order that New Operator may have full opportunity to make such reasonable investigations of the Assets or the Facility as it shall desire to make. Existing Operator shall furnish such additional financial and operating data and other information in accordance with this Section 5 and as New Operator and its representatives shall from time to time reasonably request. New Operator agrees to use discretion on any site visits which shall be conducted at times agreed upon by the parties to avoid disrupting the Facility, their employees and their patients. To the extent that any resident or employee records and other relevant records of the Facility used or developed in connection with the business conducted at such Facility are maintained by or in any computer equipment or software owned by Existing Operator, Existing Operator shall work with New Operator prior to the Closing to electronically copy the information stored in such computer equipment or software so that New Operator shall have access to the information in usable form at the Closing. To the extent Existing Operator and New Operator are unable to copy any electronic information so that New Operator has access to such information in usable form at the Closing, Existing Operator agrees to, and to cause its information technology service provider, PinComputing Company, LP, and any related information technology vendors, to reasonably cooperate following the Closing to complete the transfer of such electronic information.

- (b) From the Effective Date up through and including the date that is thirty (30) days from the Effective Date (the "Due Diligence Period"), if Existing Operator determines, in its sole and absolute discretion, that it does not desire to accept the Assets or the operations of the Facilities, New Operator shall have the right to give written notice to Existing Operator electing to terminate this Agreement, provided such notice is delivered to Existing Operator prior to 11:59 p.m. on the last day of the Due Diligence Period. The foregoing provision shall not restrict the New Operator from performing further due diligence relating to the Facilities at any time prior to the Closing Date.
- (c) All such inspections and investigations by New Operator and its authorized representatives described above shall be completed at New Operator's risk without any liability to Existing Operator, except to the extent caused by Existing Operator's gross negligence or willful misconduct. Subject to the foregoing sentence, in addition to any other indemnification obligations set forth under this Agreement, New Operator hereby agrees to indemnify and hold harmless Existing Operator against any damages or injuries associated with completion of such inspections or investigations, which undertaking shall survive the termination of this Agreement or the conveyance of the Assets by Existing Operator to New Operator, as applicable, for the period of six (6) months.
- 5.4 <u>Satisfaction of Conditions</u>. Existing Operator shall use its commercially reasonable efforts to satisfy any condition set forth in Section 1.7 or Section 1.8 that is within its power and control to satisfy, including, without limitation using commercially reasonable effort to obtain, on or prior to the Closing, all consents, if any, necessary for Existing Operator to fulfill its obligations to consummate the transactions contemplated hereby.
- 5.5 <u>Compliance with Laws.</u> Existing Operator shall use its commercially reasonable efforts to comply in all material respects with all applicable Laws of any Governmental Authority, in conjunction with the execution, delivery and performance of this Agreement and the transactions contemplated hereby.

- 5.6 <u>Further Documentation</u>. Existing Operator agrees that, for a reasonable period of time following the Closing, upon request by New Operator, it will do, execute, acknowledge, and deliver, or cause to be done, executed, acknowledged, and delivered, all such further acts, deeds, assignments, transfers, conveyances and assurances as may be reasonably required, without enlarging or extending any liability of Existing Operator beyond what is otherwise contemplated by this Agreement, to comply with its obligations hereunder.
- 5.7 <u>Taxes</u>. Existing Operator shall file federal, state, local, and, to the extent applicable, estimates and reports and pay all amounts then due, for all periods through and including the Closing Date to the extent necessary to transfer the Facility to New Operator in accordance with the terms of this Agreement.
- 5.8 <u>No Disposition of Assets</u>. Existing Operator shall not sell, lease or otherwise dispose of or distribute any of the Assets or properties related thereto or necessary for operation of any Facility other than obsolete Assets or in the ordinary course of business or as otherwise permitted by this Agreement.
- Confidentiality. Existing Operator shall use its commercially reasonable efforts to keep confidential all information relating to the terms of this Agreement and all information relating to New Operator, its officers and directors (other than information which is a matter of public knowledge or which has heretofore been or is hereafter published in any publication for public distribution or filed as public information with any Governmental Authority or disclosed pursuant to an order, subpoena or demand of any Governmental Authority or as is necessary to be disclosed to lenders, Governmental Authorities, Existing Operator and its representatives and third parties in order to consummate the transaction contemplated herein, including, without limitation, any information that is included in any filings with the Court in connection with the Bankruptcy Action) and such information shall not at any time be used for the advantage of, or disclosed to third parties (including employees at the Facility) by Existing Operator or its representatives, to the detriment of Existing Operator or its officers and directors. This provision may be enforced by temporary and permanent injunction in addition to any other available remedies, it being understood that a breach of this provision will cause Existing Operator irreparable harm.
- No Solicitation. Existing Operator agrees, it shall not, after the Effective Date and before the Closing Date, enter into any binding contract, option or other agreement relating to any acquisition of all or a substantial portion of the assets of, or any equity interest in Existing Operator or any business combination involving Existing Operator.
- 5.11 <u>Survey Reports.</u> Existing Operator shall provide to New Operator, promptly after receipt of the same, any Survey Reports including, without limitation, reports, waivers of deficiency, plans of correction, and any other investigation reports issued with respect to the Facility between the Effective Date and the Closing Date and received by Existing Operator in the ordinary course. Existing Operator shall notify New Operator upon the occurrence of any survey or investigation of the Facility, regardless of whether the applicable Health Care Authority has issued a formal report or finding (such notice may be done verbally to Natalie Holland, or such other representative as New Operator may designate from time to time, by telephone if promptly followed by written notification).

5.12 <u>Supplemental Information</u>.

(a) From time to time prior to the Closing, Existing Operator will promptly disclose in writing to New Operator any matter hereafter arising within Existing Operator's Knowledge which, if existing, occurring or known as of the Effective Date would have been required to be disclosed to New Operator or which would render inaccurate any of the representations, warranties or statements set forth

in Section 4 hereof. No information provided to a party pursuant to this Section shall automatically constitute a breach of this Agreement nor shall it be deemed to cure any breach of any representation, warranty or covenant made in this Agreement.

- Existing Operator shall, from time to time prior to or at the Closing, by notice in accordance with the terms of this Agreement (provided that Existing Operator may deliver such notice by electronic mail), supplement or amend any schedules referred to in Section 4 of this Agreement or referenced herein with the consent of the New Operator as set forth below; provided that if any such notice is delivered within two (2) Business Days prior to the date scheduled for the Closing, then New Operator may, at its option, extend the date of the Closing to a date not later than five (5) Business Days after such notice is delivered and Existing Operator provides to New Operator all information possessed by Existing Operator regarding the matters identified in such supplement or amendment. No revised schedule shall become effective until approved by New Operator. The updating and amendment of any schedule shall not cure or constitute a waiver of any breach of any representation, warranty or covenant that occurred prior to the date of such amendment unless expressly consented to in writing by New If New Operator does not consent to such amendment and thus Existing Operator's representations and warranties were true and correct as of the Effective Date, but are not true and correct as of the Closing Date, and Existing Operator therefore determines that it cannot deliver the certificate required by Section 1.5(i), New Operator's sole remedy shall be to terminate this Agreement in accordance with Section 8.1(d), and Existing Operator shall have no further liability hereunder, so long as the Existing Operator is not otherwise in breach of this Agreement. Likewise if New Operator does not consent as set forth in this Section 5.12(b), and thus Existing Operator's representations and warranties were true and correct as of the Effective Date, but are not true as a result of such supplement or amendment, Existing Operator shall be entitled to terminate this Agreement in accordance with Section 8.1(e) and neither party shall have any further liability hereunder, provided that such party is not otherwise in breach of this Agreement. Notwithstanding anything to the contrary in this Section 5.12(b), to the extent that any update or amendment of any schedule by Existing Operator is necessary as a result of Existing Operator's operation of the applicable Facility in the ordinary course consistent with past practices, New Operator's consent to or approval of such updated or amended schedule shall not be unreasonably withheld, conditioned or delayed.
- (c) Upon request, Existing Operator shall deliver to New Operator within fifteen (15) Business Days after the end of each calendar month following the Effective Date and up to the Closing Date the unaudited operating statement of the Facility for such calendar month.
- (d) Upon request, Existing Operator shall deliver to New Operator within fifteen (15) days after the end of each calendar month following the Effective Date and up to the Closing Date updated Resident Census Information for the Facility relating to such calendar month.
- (e) Upon written request, Existing Operator shall deliver to New Operator any and all cost reports filed with respect to the Facility from and after the Effective Date through the Closing Date.
- (f) Existing Operator shall notify New Operator promptly of its receipt of any written notice of the presence, release or threat of release or placement on, in or from the Facility, or the generation, transportation, storage, treatment, or disposal at the Facility, of any Hazardous Substance.
- (g) Existing Operator shall give New Operator prompt written notice of any actual or any threatened or contemplated condemnation of any part of the Facility of which it receives notice.
- 5.13 <u>Assumed Contracts</u>. Existing Operator shall give notice to all parties under the Assumed Contracts in order to obtain from such parties any consent required to be obtained to assign such Assumed

Contract to the New Operator on or prior to the Closing Date. Existing Operator shall be responsible for all costs and expenses under such Assumed Contracts prior to the Closing Date. The "Assumed Contracts" shall include all contracts and leases other than the New Operator Excluded Contracts.

SECTION 6 COVENANTS OF NEW OPERATOR

New Operator covenants and agrees with Existing Operator that:

- 6.1 New Operator shall use its commercially reasonable efforts to keep Confidentiality. confidential all information relating to the terms of this Agreement, all information relating to the Assets, Existing Operator, its officers and directors, and all financial statements, drawings, designs, customer and supplier lists relating to Existing Operator received by it (other than information which is a matter of public knowledge or which has heretofore been or is hereafter published in any publication for public distribution or filed as public information with any Governmental Authority or disclosed pursuant to an order, subpoena or demand of any Governmental Authority or as is necessary to be disclosed to lenders, Governmental Authorities, New Operator, and their respective representatives and third parties in order to consummate this transaction, including, without limitation, any information that is included in any filings with the Court in connection with the Bankruptcy Action) and such information shall not at any time be used for the advantage of, or disclosed to third parties (including employees at the Facility) by New Operator or its representatives to the detriment of Existing Operator or its officers and directors. This provision may be enforced by temporary and permanent injunction in addition to any other available remedies, it being understood that a breach of this provision will cause Existing Operator irreparable harm.
- 6.2 <u>Compliance with Laws.</u> New Operator shall comply in all material respects with all applicable Laws of any Governmental Authority, in conjunction with the execution, delivery and performance of this Agreement and the transactions contemplated hereby.
- Assumption of and Compliance with the Contracts. New Operator shall review the contracts and leases during the Due Diligence Period and, on or prior to the expiration of the Due Diligence Period, shall give notice to Existing Operator indicating which of the contracts and leases that New Operator shall not assume at Closing (the "New Operator Excluded Contracts").
- Consents, Etc. New Operator shall take reasonable steps and make all filings required by governmental bodies and other regulatory authorities, and use all reasonable efforts to obtain all Health Care Licenses and other permits, approvals, authorizations, certifications, and consents of all third parties required to consummate the transactions contemplated by this Agreement. Without limitation, promptly as practicable after the expiration of the Due Diligence Period (provided that Existing Operator cooperates with New Operator in the filing of such applications and provides, in a timely manner, all Existing Operator documents to New Operator necessary for New Operator to file such applications), New Operator shall file applications to obtain the requisite licenses to operate the Facility. Each party shall furnish promptly to each other all information that is not otherwise available to the other party and that such party may reasonably request in connection with any such filing.
- 6.5 <u>Satisfaction of Conditions</u>. New Operator shall use its commercially reasonable efforts to satisfy any condition set forth in Section 1.7 or Section 1.8 that is within its power and control to satisfy, including, without limitation using commercially reasonable effort to obtain, on or prior to the Closing, all consents, if any, necessary for New Operator to fulfill its obligations to consummate the transactions contemplated hereby.

SECTION 7 INDEMNIFICATION

- Agreement to Defend. In the event any action, suit, proceeding or investigation of the nature specified in Section 7.2, Section 7.3, or Section 7.4 hereof is commenced, whether before or after a Closing, all of the parties hereto agree to cooperate and use their best efforts to defend against and respond thereto. No party shall have any obligation to indemnify the other party with respect to any claim for indemnification arising out of or based on fraud, knowing and intentional misrepresentations or knowing and intentional breaches by the party seeking indemnification or any of its officers, employees, directors or agents.
- 7.2 <u>Indemnification by Existing Operator</u>. Existing Operator shall, jointly and severally, indemnify, protect, defend, exculpate and hold New Operator and its Affiliates, members, partners, directors, shareholders, officers, employees and agents (collectively, the "New Operator Indemnified Parties") harmless from and against, and agree promptly to defend New Operator Indemnified Parties from and reimburse New Operator Indemnified Parties for, any and all Losses (the "New Operator Indemnified Losses") which New Operator Indemnified Parties may at any time suffer or incur, or become subject to, as a result of or in connection with:
- (a) Any and all obligations of Existing Operator of any nature whatsoever, including, without limitation, all liabilities and obligations with respect to claims, damages, suits, proceedings or injury, related to or arising out of Existing Operator's ownership or operation of the Assets or the Facility prior to Closing, whether such obligation accrues before or after the Closing Date, except the Assumed Liabilities;
- (b) Any breach or inaccuracy of any of the representations or warranties made by Existing Operator in or pursuant to this Agreement or in any instrument, certificate or affidavit delivered by Existing Operator at the Closing, or from any misrepresentation in or omission from this Agreement or any Exhibit, Schedule, certificate, or other executed document furnished or to be furnished to New Operator hereunder;
- (c) Any breach by Existing Operator of its covenants, agreements, undertakings, liabilities or obligations under this Agreement or in any instrument, certificate or affidavit delivered by Existing Operator at the Closing;
- (d) Any general liability or professional liability claim with respect to the Facility related to or arising out of Existing Operator's ownership of the Assets or operation of the Facility prior to Closing, whether such obligation accrues before or after the Closing Date; and
- (e) Any and all claims, including any suit, action, or other proceeding brought by applicable Governmental Authorities or other third party payors against New Operator, the Assets or the Facility (A) arising from the ownership and operation thereof by Existing Operator prior to Closing or (B) as to any overpayments or other claims made to Existing Operator by third parties including, without limitation, overpayments made with respect to Medicaid, Medicare and any other third-party payor program.
- 7.3 Indemnification by New Operator. New Operator shall indemnify, protect, defend, exculpate

and hold Existing Operator, and its members, partners, directors, officers, employees and agents (collectively, the "Existing Operator Indemnified Parties") harmless from and against, and agree promptly to defend Existing Operator Indemnified Parties from and reimburse Existing Operator Indemnified Parties for, any and all Losses (the "Existing Operator Indemnified Losses") which Existing Operator Indemnified Parties may at any time suffer or incur, or become subject to, as a result of or in connection with:

- (a) Any and all obligations of New Operator of any nature whatsoever, including, without limitation, all liabilities and obligations with respect to claims, damages, suits, proceedings or injury, related to or arising out of the ownership of the Assets or operation of the Facility after Closing, whether such obligation accrues before or after the Closing Date, except such obligations as may be expressly assumed or retained by Existing Operator herein;
- (b) Any breach or inaccuracy of any of the representations or warranties made by New Operator in or pursuant to this Agreement or in any instrument, certificate or affidavit delivered by New Operator at the Closing, or from any misrepresentation in or omission from this Agreement or any Exhibit, Schedule, certificate, or other executed document furnished or to be furnished to Existing Operator hereunder;
- (c) Any failure by New Operator to carry out, perform, satisfy and discharge those Assumed Liabilities assumed by New Operator, or any of its other covenants, agreements, undertakings, liabilities or obligations under this Agreement or in any instrument, certificate or affidavit delivered by New Operator at the Closing; and
- (d) Any and all claims, including any suit, action, or other proceeding brought by any Governmental Authority against Existing Operator arising from the ownership and operation of the Facility by New Operator after Closing.

7.4 Notification of Claims.

- (a) A party entitled to be indemnified pursuant to Section 7.2 or 7.3 (the "Indemnified Party") shall notify the party liable for such indemnification (the "Indemnifying Party") in writing of any claim or demand which the Indemnified Party has determined has given or could give rise to a right of indemnification under this Agreement, as soon as possible after the Indemnified Party becomes aware of such claim or demand; provided, however, that the Indemnified Party's failure to give such notice to the Indemnifying Party in a timely fashion shall not result in the loss of the Indemnified Party's rights with respect thereto except to the extent the Indemnified Party is prejudiced by the delay. Subject to the Indemnifying Party's right to defend in good faith third party claims as hereinafter provided, the Indemnifying Party shall satisfy its obligations under this Section 5 within thirty (30) days after the receipt of written notice thereon from the Indemnified Party, it being agreed that the Indemnifying Party need not satisfy such obligations during any period in which the Indemnifying Party is defending in good faith an applicable third party claim in the manner described herein below.
- (b) If the Indemnified Party shall notify the Indemnifying Party of any claim or demand pursuant to Section 7.4(a), and if such claim or demand relates to a claim or demand asserted by a third party against the Indemnified Party which the Indemnifying Party acknowledges is a claim or demand for which it must indemnify or hold harmless the Indemnified Party under Section 7.2 or 7.3, the Indemnifying Party shall have the right to either (i) pay such claim or demand or (ii) employ counsel reasonably acceptable to the Indemnified Party to defend any such claim or demand asserted against the Indemnified Party. The Indemnified Party shall have the right to participate in the defense of any such claim or demand at its own expense. The Indemnifying Party shall notify the Indemnified Party in

writing, as promptly as possible (but in any case before the due date for the answer or response to a claim) after the date of the notice of claim given by the Indemnified Party to the Indemnifying Party under Section 7.4(a) of its election to defend in good faith any such third party claim or demand. So long as the Indemnifying Party is defending in good faith any such claim or demand asserted by a third party against the Indemnified Party, the Indemnified Party shall not settle or compromise such claim or demand. The Indemnified Party shall make available to the Indemnifying Party or its agents all records and other materials in the Indemnified Party's possession reasonably required by it for its use in contesting any third party claim or demand. The Indemnified Party shall have no obligation to defend any such claim or demand regardless of whether the Indemnifying Party elects to do so.

- (c) No Indemnified Party may settle or compromise any claim or consent to the entry of any judgment with respect to which indemnification is being sought hereunder without the prior written consent of the Indemnifying Party, unless (i) the Indemnifying Party fails to assume and maintain the defense of such claim pursuant to Section 7.4(b) or (ii) such settlement, compromise or consent includes an unconditional release of the Indemnifying Party from all liability arising out of such claim and does not contain any equitable order, judgment or term which in any manner affects, restrains or interferes with the business of the Indemnifying Party or any Affiliate of the Indemnifying Party. An Indemnifying Party may not, without the prior written consent of the Indemnified Party, settle or compromise any claim or consent to the entry of any judgment with respect to which indemnification is being sought hereunder unless such settlement, compromise or consent includes an unconditional release of the Indemnified Party from all liability arising out of such claim and does not contain any equitable order, judgment or term that in any manner affects, restrains or interferes with the business of the Indemnified Party or any of the Indemnified Party's Affiliates.
- If the Existing Operator Indemnified Parties desire to seek indemnification under (d) Section 7.3 with respect to any breach of a representation or warranty or covenant or claim, suit or demand that the Existing Operator Indemnified Parties believe will give rise to an indemnification claim (an "Existing Operator Claim"), the Existing Operator Indemnified Parties shall give notice of such Existing Operator Claim to New Operator (an "Existing Operator Claim Notice"), which notice shall describe in reasonable detail the nature of the Existing Operator Claim and the amount of such Existing Operator Claim (the "Existing Operator Claim Amount"). New Operator shall respond to any Existing Operator Claim Notice (an "Existing Operator Claim Response") within twenty (20) Business Days (the "Existing Operator Claim Response Period") after the date that the Existing Operator Claim Notice is received. Any Existing Operator Claim Notice or Existing Operator Claim Response shall be given in accordance with the notice requirements hereunder, and any Existing Operator Claim Response shall specify whether or not the Existing Operator Claim described in the related Existing Operator Claim Notice is disputed (an "Existing Operator Claim Dispute Notice"). If New Operator fails to give an Existing Operator Claim Response within the Existing Operator Claim Response Period or does not dispute the Existing Operator Claim described in an Existing Operator Claim Notice, then the Existing Operator Claim shall be deemed to be accepted and, if New Operator does not pay the Existing Operator Claim prior to the end of the Existing Operator Claim Response Period, the Existing Operator Indemnified Parties may pursue whatever legal remedies may be available to recover the Existing Operator Claim Amount as to which the Existing Operator Indemnified Parties are seeking indemnification.
- (e) In the event that any Existing Operator Claim is subject to an Existing Operator Claim Dispute Notice such Existing Operator Claim shall become an "Unresolved Existing Operator Claim". New Operator and the Existing Operator Indemnified Parties shall attempt in good faith to mutually agree upon the validity and amount of the Unresolved Existing Operator Claim pursuant to discussions between senior representatives of the parties who have authority to settle the same. Such discussions shall commence between the parties not more than ten (10) Business Days following delivery

of the Existing Operator Claim Dispute Notice and shall not last for more than ten (10) Business Days following the date of commencement of such discussions. In the event that the parties mutually agree upon the amount of the Unresolved Existing Operator Claim pursuant to such discussions (the "Resolved Existing Operator Claim") and the New Operator does not pay the amount of the Resolved Existing Operator Claim to the Existing Operator Indemnified Parties within five (5) Business Days thereafter, the Existing Operator Indemnified Parties may pursue whatever legal remedies may be available to recover the amount of the Resolved Existing Operator Claim. If through negotiation the parties are unable to agree upon the validity and amount of the Unresolved Existing Operator Claim within thirty (30) calendar days after the commencement of the negotiations, the dispute shall be referred to litigation in accordance with Section 7.11. In the event that a final and non-appealable judgment of a court of competent jurisdiction upholds all or a portion of the amount of an Unresolved Existing Operator Claim (the "Existing Operator Judgment Claim") and the New Operator does not pay the amount of the Existing Operator Judgment Claim to the Existing Operator Indemnified Parties within five (5) Business Days after the judgment becomes final and non-appealable, the Existing Operator Indemnified Parties may pursue whatever legal and/or equitable remedies may be available to recover such Existing Operator Judgment Claim amount. In the event that an Unresolved Existing Operator Claim is referred to litigation in accordance with Section 7.11, the losing party in such litigation shall pay all expenses connected with such litigation.

- If the New Operator Indemnified Parties desire to seek indemnification under Section 7.2 with respect to any breach of a representation or warranty or covenant or claim, suit or demand that the New Operator believes will give rise to an indemnification claim (a "New Operator Claim"), the New Operator Indemnified Parties shall give notice of such New Operator Claim to the Existing Operator (a "New Operator Claim Notice"), which notice shall describe in reasonable detail the nature of the New Operator Claim and the amount of such New Operator Claim (the "New Operator Claim Amount"). The Existing Operator shall respond to any New Operator Claim Notice (a "New Operator Claim Response") within twenty (20) Business Days (the "New Operator Claim Response Period") after the date that the New Operator Claim Notice is received. Any New Operator Claim Notice or New Operator Claim Response shall be given in accordance with the notice requirements hereunder, and any New Operator Claim Response shall specify whether or not the New Operator Claim described in the related New Operator Claim Notice is disputed (a "New Operator Claim Dispute Notice"). If the Existing Operator fails to give a New Operator Claim Response within the New Operator Claim Response Period or does not dispute the New Operator Claim described in a New Operator Claim Notice, then the New Operator Claim shall be deemed to be accepted and the New Operator may pursue whatever legal and/or equitable remedies may be available to recover the New Operator Claim Amount as to which the New Operator is seeking indemnification.
- In the event that any New Operator Claim is subject to a New Operator Claim Dispute Notice such New Operator Claim shall become an "Unresolved New Operator Claim". The New Operator Indemnified Parties and the Existing Operator shall attempt in good faith to mutually agree upon the validity and amount of the Unresolved New Operator Claim pursuant to discussions between senior representatives of the parties who have authority to settle the same. Such discussions shall commence between the parties not more than ten (10) Business Days following delivery of the New Operator Claim Dispute Notice and shall not last for more than ten (10) Business Days following the date of commencement of such discussions. In the event that the parties mutually agree upon the amount of the New Operator Claim pursuant to such discussions, the Existing Operator shall pay the New Operator Indemnified Parties the amount agreed. If the matter is not resolved directly through negotiation within thirty (30) calendar days after the commencement of the negotiations, the dispute shall be referred to litigation in accordance with Section 9.11. In the event that the court upholds all or a portion of the New Operator Claim, the Existing Operator shall be obliged to pay the amount of such New Operator Claim as is directed by the court. In the event that the validity of a New Operator Claim is referred to the court, the

losing party in such litigation shall pay all expenses connected with such litigation.

- 7.5 <u>Survival</u>. All of the representations and warranties of the parties contained in this Agreement shall survive the Closing and continue in full force and effect for a period of eighteen (18) months following the Closing; provided that the representations and warranties set forth in Sections 4.1, 4.2, 4.4, 4.6 or 4.19 shall survive until ninety (90) days after the expiration of the applicable statute of limitations with respect to the matters covered thereunder.
- 7.6 Exclusive Remedy. Subject to Section 8 and any rights to obtain injunctive relief or specific performance, the parties acknowledge and agree that their sole and exclusive remedy with respect to any and all claims (other than claims arising out of or based on fraud, knowing and intentional misrepresentations or knowing and intentional breaches by the party seeking indemnification or any of its officers, employees, directors or agents) for any breach of any representation, warranty, covenant, agreement or obligation set forth herein or otherwise relating to the subject matter of this Agreement shall be pursuant to the indemnification provisions set forth in this Section 7.

SECTION 8 TERMINATION AND ABANDONMENT

- 8.1 <u>Method of Termination</u>. This Agreement may be terminated and the transactions herein contemplated may be abandoned at any time on or before the Closing:
 - (a) by mutual written consent of all of the parties hereto;
- (b) by Existing Operator, if New Operator shall have materially breached or failed to perform any of its representations, warranties, covenants or agreements set forth in this Agreement, which breach or failure is incapable of being cured, or is not cured, by New Operator within thirty (30) calendar days following receipt of written notice from the Existing Operator of such breach or failure; provided that Existing Operator agrees that Existing Operator shall have no right to terminate, or permit the termination of, this Agreement without the prior written consent of Landlord (which consent may be withheld in its sole discretion) and that any such purported termination without the consent of Landlord shall be void and of no force or effect;
- (c) by the New Operator, if the Existing Operator shall have materially breached or failed to perform any of its representations, warranties, covenants or agreements set forth in this Agreement, which breach or failure is incapable of being cured, or is not cured, by the Existing Operator within thirty (30) calendar days following receipt of written notice from the New Operator of such breach or failure;
 - (d) by New Operator pursuant to Sections 2.14(a), 5.4(b) or 5.12 hereof; or
 - (e) by Existing Operator pursuant to Section 5.12.
- 8.2 <u>Procedure Upon Termination</u>. In the event of termination and abandonment pursuant to Section 8.1 hereof, this Agreement shall terminate and shall be abandoned, without further action by any of the parties hereto. If this Agreement is terminated for any reason:
- (a) Each of the parties will return all documents and other material of any other party relating to the transactions contemplated hereby to the party furnishing the same. New Operator shall destroy all electronic copies of Existing Operator's resident records, employee records or other relevant records made by or provided to New Operator pursuant to this Agreement. All information received by

any party hereto with respect to the business of any other party (other than information which is a matter of public knowledge or which has heretofore been or is hereafter published in any publication for public distribution or filed as public information with any Governmental Authority, including, without limitation, any information that is included in any filings with the Court in connection with the Bankruptcy Action) shall be kept confidential and shall not at any time be used for the advantage of, or disclosed to third parties by, such party to the detriment of the party furnishing such information; and

(b) Other than on the basis of a breach of this Agreement prior to its termination, no party hereto shall have any liability or further obligation to any other party to this Agreement other than under the provisions of Sections 5.3, 5.9 and 6.1 hereof.

8.3 Remedies for Default.

- (a) If this Agreement is terminated pursuant to Section 8.1, all obligations of the parties hereunder shall terminate, except for the obligations set forth or referenced in this Section 8, which shall survive the termination of this Agreement, and except that no such termination shall relieve any party from liability for any prior willful breach of this Agreement. If any of the conditions set forth in Section 1.7 or Section 1.8 of this Agreement have not been satisfied, the party for whose benefit such provisions exist may nevertheless elect to proceed with the consummation of the transactions contemplated hereby. If the conditions of Section 1.8 have been satisfied and Existing Operator shall not complete the Closing, and New Operator is not in default hereunder, then New Operator's sole remedies at law or in equity shall be to terminate this Agreement, after which the parties shall have no further rights hereunder, other than under the provisions of Sections 5.3, 5.9 and 6.1 hereof.
- (b) In the event that after the expiration of the Due Diligence Period, without termination of this Agreement by New Operator, New Operator shall not complete the Closing, and Existing Operator is not in default hereunder, then Existing Operator's sole remedy shall be to terminate this Agreement, after which the parties shall have no further rights hereunder, other than under the provisions of $\underline{\text{Sections}}$ $\underline{5.3}$, $\underline{5.9}$ and $\underline{6.1}$ hereof.

SECTION 9 MISCELLANEOUS

- 9.1 <u>Further Assurances</u>. Each of the parties hereto agrees to execute and deliver any and all further agreements, documents or instruments reasonably necessary to effectuate this Agreement and the transactions referred to herein, contemplated hereby or reasonably requested by the other party to perfect or evidence their rights hereunder.
- Notices. All notices, requests, demands and other communications required or permitted hereunder shall be in writing and shall be personally delivered, or sent by facsimile transmission (provided a copy is thereafter promptly mailed as hereinafter provided), or sent by overnight commercial delivery service (provided a receipt is available with respect to such delivery), or mailed by first-class registered or certified mail, return receipt requested, postage prepaid (and shall be effective when received, if sent by personal delivery or by facsimile transmission or by overnight delivery service, or on the third (3rd) day after mailing, if mailed):

If to Existing Operator, to:

Bloomfield Health Facilities, L.P. 5420 W. Plano Parkway Plano, Texas 75093

Attn: Thomas D. Scott Email: tom.scott@pcitexas.net

Fax: 972-931-3801

with a copy to (which shall not constitute notice):

Robert J. Riek, General Counsel 5500 W. Plano Parkway, Suite 210 Plano, Texas 75093 Email: bob@rwtx.com

Fax: 972-767-6235

If to New Operator:

c/o Genesis Healthcare, Inc. 101 East State Street Kennett Square, Pennsylvania 19348

Attn: Law Department

Email: Michael.Sherman@GenesisHCC.com

Fax: (484) 733-5449

with a copy (which shall not constitute notice) to:

Williams Mullen 200 South 10th Street, Suite 1600 Richmond, VA 23219 Attention: Beth G. Hungate-Noland

Fax: (804) 420-6507

Email: bhungate-noland@williamsmullen.com

or to such other person or address as any party hereto shall furnish to the other parties hereto in writing pursuant to this Section 9.2.

- 9.3 <u>Payment of Expenses.</u> In the event of any dispute or controversy arising out of this Agreement, including in connection with the interpretation of any term or condition of this Agreement, the prevailing party shall recover from the non-prevailing party all reasonable costs and expenses, including reasonable attorneys' fees, incurred by the prevailing party.
- Entire Agreement; Amendment; Waiver. This Agreement together with the other agreements referred to herein, constitutes the entire understanding between the parties with respect to the subject matter hereof, superseding all negotiations, prior discussions and preliminary agreements. This Agreement may not be modified or amended except in writing signed by the parties hereto. No waiver of any term, provision or condition of this Agreement, in any one or more instances, shall be deemed to be or be construed as a further or continuing waiver of any such term, provision or condition or as a waiver of any other term, provision or condition of this Agreement. No failure to act shall be construed as a waiver of any term, provision, condition or rights granted hereunder.
- 9.5 <u>Assignment</u>. Neither this Agreement nor the rights, duties or obligations arising hereunder shall be assignable or delegable by either party hereto without the express prior written consent of the other party hereto.

- 9.6 <u>Joint Venture; Third Party Beneficiaries</u>. Nothing contained herein shall be construed as forming a joint venture or partnership between the parties hereto with respect to the subject matter hereof. The parties hereto do not intend that any third party shall have any rights under this Agreement.
- 9.7 <u>Representation By Counsel</u>. The parties hereto acknowledge that they have been represented by independent legal counsel of their choosing throughout all of the negotiations which preceded the execution of this Agreement, and that each party has executed this Agreement with the consent and on the advice of such independent legal counsel. This Agreement is a negotiated document. As a result, any rule of construction providing for any ambiguity in the terms of this Agreement to be construed against the draftsperson of this Agreement shall be inapplicable to the interpretation of this Agreement.
- 9.8 <u>Captions</u>. The section headings contained herein are for convenience only and shall not be considered or referred to in resolving questions of interpretation.
- 9.9 <u>Counterparts</u>. This Agreement may be executed and delivered (including by facsimile transmittal, portable document format, or by similar electronic means, which for purposes of this Agreement shall be deemed to be an original signature) in one or more counterparts and all such counterparts taken together shall constitute a single original Agreement.
- 9.10 <u>Governing Law</u>. This Agreement shall be governed by the laws of the State of New Mexico as to, including, but not limited to, matters of validity, construction, effect and performance but exclusive of its conflicts of laws provisions.
- 9.11 <u>Jurisdiction</u>. Each party hereto consents to the jurisdiction of the courts of the State of New Mexico, or if it can acquire jurisdiction, in the United States District Court for the District of Delaware as to claims arising under or brought in connection with this Agreement and the transactions contemplated herein.
- 9.12 <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HEREBY KNOWINGLY, VOLUNTARILY AND INTENTIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY RIGHT TO TRIAL BY JURY IN ANY LEGAL ACTION BROUGHT ON OR WITH RESPECT TO THIS AGREEMENT, INCLUDING TO ENFORCE OR DEFEND ANY RIGHTS HEREUNDER, AND AGREES THAT ANY SUCH ACTION SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.
- 9.13 <u>Calculation of Time Periods</u>. Unless otherwise specified, in computing any period of time described herein, the day of the act or event on which the designated period of time begins to run shall not be included and the last day of the period so computed shall be included, unless such last day is a Saturday, Sunday or legal holiday, in which event the period shall run until the next day which is not a Saturday, Sunday or a legal holiday. Further, unless otherwise specified, any reference to a specified number of days shall be deemed to refer to calendar days.

[SIGNATURES BEGIN ON FOLLOWING PAGE]

[Signature Page to Operations Transfer Agreement]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the day and year first above written.

EXISTING OPERATOR: Bloomfield Health Facilities, L.P.

a Texas limited partnership

By: Bloomfield Health Facilities GP, LLC its general partner

Title: Manager

[Signatures Continue on Following Page]

[Signature Page to Operations Transfer Agreement]

NEW OPERATOR:	a []],
	By:	
	Title:	

EXHIBIT A

Defined Terms

"Affiliate" means, as applied to any Person, any other Person directly or indirectly controlling, controlled by or under common control with, that Person. For the purposes of this definition, "control" (including with correlative meanings, the terms "controlling", "controlled by", and "under common control with") as applied to any Person, means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of that Person, whether through ownership of voting securities or by contract or otherwise.

"Bankruptcy Action" shall mean the action pending in the Court styled as *In re Preferred Care Inc. et. al, Case No. 17-44642* and any other bankruptcy proceeding filed by or against Existing Operator.

"Business Day" shall mean any day other than a Saturday, Sunday or holiday on which national banking associations in the State of New York are authorized or required to be closed.

"Court" shall mean United States Bankruptcy Court for the Northern District of Texas, Fort Worth Division and any other court in which a Bankruptcy Action arises following the Effective Date.

"Environmental Law" shall mean each and every applicable federal, state, local and foreign law, statute, ordinance, regulation, rule, judicial or administrative order or decree, permit, license, approval, authorization or similar requirement of each and every federal, state, local and foreign governmental agency or other Governmental Authority, having or claiming to have authority with respect to human health and safety or the environment affecting the Facility, including, but not limited to, the Resource Conservation and Recovery Act (42 U.S.C. Section 6901 *et seq.*) as amended, the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. Section 9601 *et seq.*) as amended, the Clean Water Act (33 U.S.C. Section 1251 *et seq.*) as amended.

"Environmental Permit" shall mean any permit, license, approval, consent or other authorization required by or pursuant to any applicable Environmental Law.

"ERISA Affiliate" shall mean each trade or business (whether or not incorporated) which together with any of Old Operators is or ever was treated as a single employer under Section 414(b), (c), (m), (o) or (t) of the Internal Revenue Code of 1986, as amended.

"Existing Operator's Knowledge" shall mean any fact or matter that is known or should have been known by Existing Operator or Existing Operator's officers or directors with due inquiry.

"Governmental Authority" shall mean the government of the United States or any foreign country or any state or political subdivision thereof and any entity, body or authority exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, and other quasi-governmental entities established to perform such functions.

"Hazardous Substance" shall mean any material or substance that (i) constitutes a hazardous substance, toxic substance or pollutant (as such terms are defined by or pursuant to any Environmental Law) or (ii) is regulated or controlled as a hazardous substance, toxic substance, pollutant or other regulated or controlled material, substance or matter pursuant to any Environmental Law.

"Health Care Authority" shall mean any Governmental Authority or quasi-Governmental

Authority or any agency, intermediary, board, authority or entity concerned with the ownership, operation, use or occupancy of the Facility as a skilled nursing facility, long-term acute care facility or assisted living facility, if applicable.

"Health Care Requirements" shall mean, with respect to the Facility, all federal, state, county, municipal and other governmental statutes, laws, rules, orders, regulations, ordinances, judgments, decrees and injunctions or agreements, in each case, pertaining to or concerned with the establishment, construction, ownership, operation, use or occupancy of such Facility or any part thereof as a nursing facility, long-term acute care facility, assisted living facility or other health care facility, if applicable, and all material permits, licenses and authorizations and regulations relating thereto, including all material rules, orders, regulations and decrees of and agreements with Health Care Authorities as pertaining to such Facility.

"Law" shall mean any law, statute, regulation, ordinance, rule, order, decree, judgment, consent decree, settlement agreement or governmental requirement enacted, promulgated, entered into, agreed or imposed by any Governmental Authority.

"Lien" shall mean any mortgage, lien (except for any lien for Taxes not yet due and payable), charge, restriction, pledge, security interest, option, lease or sublease, claim, right of any third party, easement, encroachment or encumbrance.

"Loss" or "Losses" shall mean any and all liabilities, losses, costs, claims, damages (excluding consequential damages), penalties and expenses (including attorneys' fees and expenses and costs of investigation and litigation). In the event any of the foregoing are indemnifiable hereunder, the terms "Losses" and "Losses" shall include reasonable attorneys' fees and documented expenses and costs of investigation and litigation incurred by the indemnified party in enforcing such indemnity.

"New Operator's Knowledge" shall mean any fact or matter that is known or should have been known by New Operator or New Operator's officers or directors after reasonable due inquiry.

"Person" shall mean any individual, corporation, proprietorship, firm, partnership, limited partnership, trust, association or other entity.

"Resident Census Information" shall mean, with respect to the Facility, a true, correct and complete schedule (provided in accordance with Health Care Requirements related to privacy) that accurately and completely sets forth the occupancy status of such Facility, the average daily rate and other charges payable with respect thereto, the class of payment or reimbursement (<u>i.e.</u>, private, third-party payor, Medicare and Medicaid), the average monthly census of such Facility, occupancy rates and any arrearages in payments.

"Survey Report" shall mean any and all survey reports, waivers of deficiencies, plans of correction and any other investigation reports issued with respect to the Facility.

"Taxes" shall mean all taxes, charges, fees, duties, levies or other assessments, including income, gross receipts, net proceeds, ad valorem, turnover, real and personal property (tangible and intangible), sales, use, franchise, excise, value added, stamp, leasing, lease, user, transfer, fuel, excess profits, occupational, interest equalization, windfall profits, severance, employee's income withholding, other withholding, unemployment and social security taxes, which are imposed by any Governmental Authority, and such term shall include any interest, penalties or additions to tax attributable thereto.

"Tax Return" shall mean any report, claim for refund, return or other information required to be supplied to a Governmental Authority in connection with any Taxes, including any schedule or attachment thereto and including any amendment thereof.

EXHIBIT B

ASSIGNMENT AND ASSUMPTION AGREEMENT

, 2018 (the

"Effective Date"), and is between Bloomfield Health Facilities, L.P., a Texas limited partnership ("Existing Operator"), and, ("New Operator").
A. Existing Operator and New Operator are parties to an Operations Transfer Agreement ("Agreement") dated as of, 2018.
B. Existing Operator has agreed to assign to New Operator all of its right, title and interest in, to, and under those contracts identified on Exhibit A attached hereto for the Facility ("Assumed Contracts"), and that New Operator assumes Transferor's obligations with respect to such Assumed Contracts for each respective Facility in accordance with the Agreement.

This Assignment and Assumption Agreement is effective as of

NOW, THEREFORE, in consideration of the premises, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged by the parties hereto, the parties hereto, intending to be bound, hereby agree as follows:

- 1. Existing Operator hereby assigns, transfers and conveys all of its right, title and interest in, to, and under the Assumed Contracts to New Operator.
- 2. New Operator hereby assumes all liabilities with respect to the Assumed Contracts, but only to the extent that such liabilities relate to the period after the Effective Date hereof.

[Execution page to follow]

EXHIBIT C

BILL OF SALE

This BILL OF SALE is made and entered into as of,	2018 between
	Health Facilities,
L.P.,, a Texas limited partnership ("Existing Operator").	
In consideration of Ten Dollars (\$10.00) and other good and valuable considera	tion, the receipt
and sufficiency of which are hereby acknowledged, Existing Operator has transferred	d, assigned and
conveyed, and by these presents does transfer, assign and convey unto New Operator as	ny and all right,
title and interest which Seller possesses in and to, all and singular, the Assets as defined in	n the Operations
Transfer Agreement ("OTA") dated as of, 2018 between Existing Open	erator and New
Operator.	

By acceptance of this Bill of Sale, New Operator acknowledges that it is acquiring the Assets strictly on an as "AS IS" basis, and New Operator acknowledges and agrees that, except for those representations and warranties set forth in the OTA, neither Existing Operator, nor its agents, contractors or representatives have made any representations, warranties, promises, covenants, agreements or guarantees of any kind or character whatsoever, whether express of implied, oral or written, past, present or future, of, as to, concerning or with respect to the nature, quality or condition of the Assets. EXISTING OPERATOR EXRPESSLY DISCLAIMS ANY IMPLIED WARRANTIES INCLUDING WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE WITH RESEPCT TO THE ASSETS.

TO HAVE AND TO HOLD, all and singular, the Assets hereby transferred, assigned and conveyed to New Operator, its successors and assigns, to and for its own use and benefit.

[SIGNATURE PAGE TO FOLLOW]

SCHEDULE 2.9

Account

EXHIBIT D